

Order Flow and the Formation of Dealer Bids: An Analysis of Information and Strategic Behavior in the Government of Canada Securities Auctions Ali Hortacsu and Samita Sareen

NOTA DI LAVORO 145.2004

DECEMBER 2004

PRA – Privatisation, Regulation, Antitrust

Ali Hortacsu, *University of Chicago* Samita Sareen, *Duke University*

This paper can be downloaded without charge at:

The Fondazione Eni Enrico Mattei Note di Lavoro Series Index: http://www.feem.it/Feem/Pub/Publications/WPapers/default.htm

Social Science Research Network Electronic Paper Collection: http://ssrn.com/abstract=624463

The opinions expressed in this paper do not necessarily reflect the position of Fondazione Eni Enrico Mattei Corso Magenta, 63, 20123 Milano (I), web site: www.feem.it, e-mail: working.papers@feem.it

Order Flow and the Formation of Dealer Bids: An Analysis of Information and Strategic Behavior in the Government of Canada Securities Auctions

Summary

Using data on Government of Canada securities auctions, this paper shows that in countries where direct access to primary issuance is restricted to government securities dealers, Order-flow" information is a key source of private information for these security dealers. Order-flow information is revealed to a security dealer through his interactions with customers, who can place bids in the auctions only through the security dealer. Since each dealer interacts with a different set of customers, they, in effect, see different portions of the market demand and supply curves, leading to differing private inferences of where the equilibrium price might.

Keywords: Treasury auctions, Behavioural finance

JEL Classification: D82, G14, L88

This paper was presented at the EuroConference on "Auctions and Market Design: Theory, Evidence and Applications", organised by Fondazione Eni Enrico Mattei and Consip and sponsored by the EU, Rome, September 23-25, 2004.

Address for correspondence:

Samita Sareen Duke University Durham NC 27708 USA E-mail: ssareen@econ.duke.edu

1 Introduction

The asymmetric information view of financial markets posits that financial market participants have access to private information regarding the value of a given security, and use their private information to their strategic advantage. A large theoretical literature is concerned with the question of how market equilibrium will be attained, and how assets will be priced, in the presence of private information and with the strategic use of this information.¹

What is the source of "private" information in financial markets, especially in the vast markets for government securities around the world? After all, it is difficult to argue that most participants in such markets are privy to much insider information regarding the "fundamentals" underlying the value of a particular government bond. Most, if not all larger players in these markets have access to the same computer screens, and financial and political news data. Hence, aside from possessing heterogeneous priors due to exogenous reasons, the forecasts they will make will differ from each other only to the extent that their forecasting technologies are different.

There is, however, one source of truly "private" information among financial market participants, identified by the market microstructure literature as "order flow." Most large players in financial markets are intermediaries, who buy and sell securities to profit from the bid-ask spread. A source of private information is their interactions with their customers – since each dealer interacts with a different set of customers, they, in effect, see different portions of the market demand and supply curves, leading to differing "private" inferences of where the equilibrium price might lie.

In this paper, we will analyze the role of "order flow" as a source of private information in financial markets, and its consequences regarding the strategic behavior of market participants using a unique data set from the Government of Canada securities auction market. In this market, securities dealers authorized by the Bank of Canada place bids in debt issues of the Government of Canada (GoC henceforth). These securities dealers bid for themselves, but they are also allowed to submit bids on behalf of their customers. Similar to other Treasury auction data sets used in previous microeconomic studies, we have access to the entire set of bids submitted for a set of securities of offerings.² A "unique" aspect of our data set is that we are able to observe which bids are submitted by the dealers for themselves and which bids are submitted for their customers. Another "unique" aspect of our data is that, along with the final set of bids, we are also able to track how dealers modify their own and their customers' bids on the Bank of Canada's bid submission database over time.

Given this extremely rich data source, we investigate the following questions:

 $^{^1\}mathrm{See}$ O'Hara (1995) and the references therein for a comprehensive account of this theoretical literature.

 $^{^2\}mathrm{Empirical}$ studies on Treasury auctions that also have access to bidder-level data are Umlauf (1993), Gordy (1994), Nyborg, Rydqvist and Sundaresan (2002), Hortacsu (2002), Fevrier, Preguet and Visser (2002).

- 1. Is there evidence for private information in this market? If there is, how does the market mechanism aggregate this information?
- 2. How much of this private information is due to "order flow"? In particular, do dealers make use of the information contained in their customers' bids, as evidenced by observed modifications in their own bids? Assuming that dealers do modify their bids on the basis of information revealed by their customers, do they make these modifications in a manner consistent with a rational, profit-maximizing agent point of view?
- 3. It is reasonable to expect that customers know that they are revealing valuable information to their dealers, and will react accordingly. Is there any evidence for strategic behavior on the part of the customers?

Our first set of results, obtained in Sections 4-6, yields that there is a large amount of heterogeneity in the information possessed by different bidders participating in these auctions. In particular, in section 5, we document that there is a large degree of dispersion across dealers bids. In section 4, we document that these dispersed bids are the result of a information gathering process that takes place in the period leading up to the auction. Specifically, we document that official bid submissions are very much concentrated within the last 15 minutes preceding the bid submission deadline. We also document that "later" bids are much better predictors of the auction outcome than "early" bids. We also find bidding activity for long-term securities (bonds) are much more concentrated within the last few minutes of the bidding window, and that this last minute information aggregation phenomenon is much more visible for bonds than for bills. This finding is consistent with the intuition that the valuation of longterm securities is subject to much more uncertainty, and hence, the incremental value, and hence the value of waiting for, new pieces of information to arrive, is larger.

Our next set of results delve deeper into the nature of this information gathering process that precedes the auction. In particular, in section 6, we find strong evidence that dealers delay their own bids until they receive the bids of their customers, who, in turn, wait until the last minutes of the auction to bid. Once again, there is a visible difference between these temporal patterns across bonds and bills: dealers' tendency to "wait" for customer bids is much more pronounced in bond auctions as opposed to bill auctions, suggesting that the information contained in customer bids is more important in bond auctions as opposed to bill auctions.

We then analyze whether dealers indeed incorporate the information contained in customer bids in their own bids. In section 6, we analyze data on modifications in dealers' bids, and, controlling for public information that might be driving these bid modifications, we find that two sources of private information drive these bid changes. The first source of private information is the information contained in modifications to customer bids, which are observed by the dealers through which these customers bid. The second source of private information is the net long or short position of a dealer, obtained throughout the bidding process, through over-the-counter when-issued transactions with customers.

These results suggest that an important source of dispersion in dealers' bids, and hence a source of private information in this market, is the information contained in customer bids. In section 5, we also document that, controlling for size, dealers with access to more customers are better able to predict market outcomes. Hence, we our results are consistent with the simple theoretical intuition that "more information is better" when making decisions under uncertainty.

Next, we focus on the "customer" side of the market. One might expect that customers, especially those with large orders, will realize the strategic value of the information contained in their bids, and react accordingly. In particular, in section 7.2, we observe that some customers spread their bids across several dealers. We also find that customers are more likely to route their bids through more than one dealer when the customer has a large quantity of security to buy (and hence is more likely to "move the market"). this effect is more pronounced for bond auctions compared with treasury bill auctions, replicating the "information gathering" hypothesis across treasury bill and bonds observed in the dealer-side of the market.

An analysis of bids that arrive after the auction submission deadline in Section 7.1, reveals that most of these are dealer bids submitted in bond auctions. Presumably customers hold-off submitting their bids as long as they can; dealers bids that incorporate these customer bids arrive after the auction submission deadline and are rejected. icr. Interestingly, we also find in Section 7.3 that some large customers who do not route their bids through multiple dealers in an auction are in long-term relationships with their dealers. Since repeated relationships with a customer provide valuable information to the dealers, it is conceivable that dealers compensate the customers for this information. By looking at the difference in the average price paid and the cutoff price, we find that customers who are in a long-term relationship in the bond sector pay a lower price in bond auctions when they submit tenders through the dealer with whom they are in a long-term relationship, compared with when they submit tenders through other dealers. Customers who have a long-term relationship in the treasury bill sector but participate in the bond auctions, pay a lower price for bonds when they submit tenders through this long-term dealer compared with when they submit tender through other dealers. This price is comparable to that paid by customers with a long-term relationship in the bond sector, for the tenders submitted through the long-term dealer. Finally, customers in a long-term relationship pay a higher price for tenders submitted through the long-term dealer compared to customers who are not in a long-term relationship. This overpayment is greater when the long-term relationship is in bonds compared with treasury bills. It is conceivable that these customers are compensated by the long-term dealer in a sector other than Government of Canada securities. Once gain, the pay-off hypotheses show that the customer-side of the primary market reflects the last minute "information gathering" hypothesis for longer-term securities relative to shorter-term securities formulated from the dealer-side in sections 4-6.

Thus, our results shed light into the workings of the "black box" surrounding the source and aggregation of private information in government securities markets, and the formation of customer-dealer relationships as a response to the exchange of valuable information across these two parties. It also points to one of the benefits of being a dealer, that of having private access to order flow information. Of course, as in any industry with fixed costs, some revenues above marginal cost are needed to sustain entry; hence, one may regard the informational advantage possessed by the dealers as this additional revenue component.³

The importance of "order-flow" as a source of private information, and the strategic incentives possessed by dealers or marketmakers in response to having access to order flow information has been recognized by a large theoretical literature in finance (see for example, O'Hara (1995) and the numerous references within). Empirical work on the role played by "order flow" in asset pricing has been conducted in various securities markets. These include foreign exchange markets (Lyons (2001) and the references within, Evans and Lyons (2002), Ito, Lyons and Melvin (1998)), equity trades (Hansch,Naik and Viswanathan (1998)) and option trades (Easley, O'Hara and Srinivas (1998)). Aside from Hansch, Naik and Viswanathan (1998), this literature has used data aggregate, market-level data to investigate the importance of aggregate measures of order-flow and trading activity on asset prices.

Empirical work that is closest to ours has been conducted by Drudi and Massa (2001), who examine the individual trading behavior of government securities dealers on the primary (auction) market and the interdealer exchange, using a detailed fixed-income transaction database from Italy. Their starkest finding is that dealers behave strategically on the interdealer market to manipulate outcomes in the auction market. Massa and Simonov (2001) use the same data source to investigate strategic trading in the interdealer exchange, and find, consistent with our findings, that information gained from secondary market trading can affect bidding behavior in the primary market. Massa and Simonov (2003) document that long-term interactions between dealers leads to the formation of "dealer reputations," which affect the informational content that counterparties ascribe to the trades originating from these dealers.

Our results complement the findings of Drudi and Massa (2001) and Massa and Simonov (2001, 2003) in delving deeper into the sources of private information and strategic behavior in government securities markets, especially in the primary market. Isolating what drives private information is important in these

³Aside from setting up the technological infrastructure to participate in these auctions, as discussed in section 2, dealers may be thought of incurring fixed costs due to the regulations they need to comply by. For example, primary dealers are subject to minimum bidding requirements to keep their status, this may be construed as a fixed cost of doing business. Sareen (2002) argues that the primary dealership system, whereby the issuer makes access to advantages conditional on the dealers satisfying obligations is a means to resolve the agency problem between the issuer and the security dealers. When the number of security dealers is small perhaps due to high entry costs on account of obligations that a dealer has to satisfy, this argument becomes especially convincing.

markets, since theoretical and policy analyses regarding the design of the auction and surrounding market rules rely very sensitively on the exact specification of the informational and strategic environment (see Binmore and Swierbinski (2003), Sareen (2003) for recent surveys of the theoretical and empirical literature on the design of Treasury auctions). In contrast to the work of Drudi, Massa and Simonov, however, our analysis focuses on the dealer-customer interaction, rather than the interactions between different dealers.

The structure of the rest of the paper is as follows: in the next section, we describe the institutional arrangement and rules of the Government of Canada securities markets. Section 3 describes our data base, and provides summary statistics regarding the composition of securities and participating firms in this market. Section 4 describes the temporal patterns in information aggregation. Section 5 describes the heterogeneity in dealer bids, and Section 6 provides an explanation for this heterogeneity through customer bids. Section 7 describes customer and dealer relationships in light of the documented information transmission between these parties. Section 8 concludes.

2 The Institution

The Bank of Canada, on behalf of the Government of Canada (GoC henceforth), issues bonds with a maturity of 30, 10, 5 and 2 years, and treasury bills with a maturity of 1 year, 3 months and 6 months.⁴ The process for issuing a "typical" GoC debt instrument links *three* markets: the when-issued market, primary market or the auction, and the secondary market. The process begins with government securities distributors (*dealers* henceforth) typically taking *short* positions in the when-issued market through forward contracts with other participants, for the yet-to-be auctioned security.⁵ Subsequently, dealers attempt to cover these short positions by buying the security from other dealers in the when-issued market, from the issuer in the primary market, and finally from another dealer in the secondary market after the auction. In all instances, being profit maximizing agents, they will attempt to buy the securities in which they have a short position at the cheapest price.⁶ Details of the three markets are given below.

The *when-issued* market precedes the auction by a week. Participants engage in forward trading for the yet-to-be auctioned security in the *when-issued* market.

Following the when-issued market is the *primary* market, where the Bank of Canada issues GOC securities through a discriminatory price auction. Potential

⁴Treasury bills are zero-coupon bonds.

 $^{^{5}}$ The Bank of Canada designates certain institutions as distributors of government securities. These institutions are obligated to buy and sell securities to individual investors.

⁶If the price at which dealers buy the securities in which they have a short position is less (greater) than the price at which they have pre-sold the security in the when-issued market, dealers make a profit (loss) ex-post.

bidders in the government securities auctions can be classified into three groups: primary dealers, other government securities distributors and customers, where "other government securities distributors" refers to government securities distributors excluding primary dealers. The *key* distinction between government securities distributors and customers is that the latter cannot bid on their own account in the auction; rather government securities distributors submit bids on behalf of customers. Thus, Government securities distributors submit bids on their account and on behalf of the customers, being "bidders" in the former and "submitters" in the latter case. The distinction between primary dealers and other government securities distributors is that the former are subject to minimum bid obligations, as both "bidder" and "submitter", in the form of a minimum quantity constraint and a binding price constraint.⁷ A customer can chose to submit bids through more than one dealer in an auction.

Bids can be submitted as *competitive* tenders and *noncompetitive* tenders, with the **focus** of this paper being on competitive tenders (tenders henceforth).⁸ Typically, a participant's tender will comprise of price-quantity pairs, and the participant's *net position* of the yet-to-be auctioned security at the point of time the tender is submitted. Net position at a point in time refers to the participant's net holdings (whether long or short) of the security being auctioned at that point of time. A customer's tender may comprise only price-quantity pairs as a customer has the option of submitting his net positions directly to the Bank of Canada instead of communicating it through the tender(s) he submits through dealers. Participants can **revise** or cancel previously submitted tenders prior to the auction deadline; there are no limits on the number of revisions that an auction participant can make.

Submitted tenders are allotted through a discriminatory price auction.

The *primary* market is followed by trading in an active resale market for the "new issue" called the *secondary* market.

Both the when-issued and the secondary markets comprise the resale market for GOC securities. The resale market can be decomposed into two: the interdealer market and the customer-dealer market. In the customer-dealer market, institutional investors (for example, pension funds, mutual funds) trade with dealers on a bilateral over-the-counter basis over the telephone, with the result of these transactions known only to the two counterparties participating in the transaction. This is in contrast to the interdealer market that operates primarily through electronic interdealer brokers, who post on an electronic screen, bid, offers, and trade outcomes communicated to him by the dealers telephonically, without revealing the identity of the dealer. While only dealers can post quotes or trade through the interdealer brokers, **both** customers and

⁷A primary dealer has to submit his *threshold* bid at a yield which is within 5 cents of the highest accepted yield. A *threshold* bid is a bid that makes the cumulative amount of the quantity bid greater than or equal to a primary dealer's lower quantity constraint.

 $^{^{8}}$ A *noncompetitive* tender comprises a quantity subject to an upper bound of \$3 million, with a participant being allowed to submit a single *noncompetitive* tender.

dealers have **viewing** access to the electronic screens of an interdealer broker via CanPX, a data service that consolidates and disseminates the trade and quote information provided by the interdealer brokers.

3 Data

Our data set captures several aspects of the primary and the resale markets. For the primary market we have data over the period October 1998 to March 2003. For the resale market we have data over the period July 4, 2001 to September 10, 2001, and February 25, 2002 to February 27, 2003. Henceforth, a GoC security will be uniquely identified by it's maturity date and coupon rate if it is a bond, and by a maturity date if it is a treasury bill. An auction will refer to the issuance of a GoC security of a specific maturity range (30, 10, 5, 2 years for bonds and 1 year, 6 months, 3 months for treasury bills), held at a specific time; thus, an auction for a GoC security will be uniquely identified by the maturity range and the date on which it is held.

3.1 Primary Market

Two aspects of the primary market data are distinguished: the hierarchical aspect of the data, and the official vs history aspect of the data.

There are three level in our hierarchical data base: auction level, tender level, bid level.

Auction Level

For each auction we have the issue amount; issue date and maturity date of the auctioned security; total amount bid; total amount allotted; cutoff yield; total bid amount at the cutoff yield; coupon rate (if the GoC security is a bond).

Tender Level

For each tender submitted in an auction, we have the following information: tender type (competitive or noncompetitive); time-stamp of the tender indicating the time at which the tender was submitted in the auction; tender status, in that it is a submitted by the participant, cancelled by the participant or rejected by the issuer;⁹ identity of the submitter of the tender; identity of the bidder of the tender; participant type of the bidder (primary dealer, other government securities distributor, customer, Bank of Canada) and submitter (primary dealer, other government securities distributor); net position amount

⁹Tenders received before the auction deadline will have a status of "submitted" or "cancelled". A participant can cancel his last submitted tender. No bid level information is available for cancelled tenders. Tenders received after the auction deadline will have a status of "pending submission" or "pending cancellation". These tenders will either be accepted or rejected by the issuer, with a tender beyond the auction deadline usually being accepted by the issuer if it is on account of transmission failure, and rejected otherwise. The former if accepted has a status of "submitted", and the latter if accepted has the status of "cancelled". All rejected tenders have the status of "rejected".

indicating a participant's net holdings of the yet-to-be auctioned security at that time,¹⁰ with the exception of net positions reported directly to the Bank of Canada by a customer;¹¹ the allotted tenders of each bidder.

Bid Level

We have the bid amount and yield pairs for each tender submitted by a participant; the maximum amount a participant can bid as a "submitter", and as a "bidder"; and the amount allotted to each participant.

In addition to the hierarchical aspect of this data, we also distinguish the official vs history aspect of the data. As mentioned before, a bidder can revise a submitted tender before the auction deadline.¹² All tender revisions and the constituent bid level revisions, made by each participant in the auction will be referred to as the history of the auction. For each auction participant, the **unique** tender and the constituent bids in this tender that are used to determine the cutoff yield and the allotment of the auctioned security is referred to as official data.¹³ Thus, the official data is a subset of the history data.

4 Temporal Patterns in Bidding Behavior and Information Aggregation

Bank of Canada securities auctions have a fixed bid submission deadline, and bidders are allowed to submit bids for a particular auction two weeks ahead of time. Since our data set includes the time stamps for each tender (official or not) submitted by the dealers, we can analyze the timing of official bid submissions.

In Figure 1, we plot the cumulative distribution function of official bid submission times for different subsamples of our data set. First, observe that bidding activity is very much concentrated within the hour before the submission deadline. Ninety percent of all competitive bids, those that specify a price as well as a quantity, are submitted in the last 20 minutes of the submission deadline. In contrast, observe that non-competitive bids, i.e. those that do not specify a price, tend to come much later than competitive bids. The median

 $^{^{10}{\}rm A}$ net position has to reported with a revised tender only if the change in the net position since the last submitted tender exceeds \$25 million.

¹¹Customers can submit their net positions to the issuer instead of revealing them to the dealer when submitting their tenders through them. This has to be done 30 minutes before the auction deadline unless there is a change in the net position by 25 million, in which case the latest net position reported is recorded. For customers who submit net positions directly to the issuer, only the last net position amount reported by a customer before the auction deadline is available; the history of revisions in the net position is not available.

 $^{^{12}}$ While the tender revision is to be done before the auction deadline, in blah % of our sample we find the revisions done after the auction deadline.

 $^{^{13}}$ An official tender is the last submitted or cancelled tender of a participant. While in blah % of the auctions, the official tender is submitted before the auction deadline, we do have cases where the official tender was submitted after the auction deadline. This could be due to transmission, as well as, strategic reasons.

competitive bid comes in 7.9 minutes before the deadline, whereas the median non-competitive bids comes in 26.5 minutes before the deadline.

A similar phenomenon of "last minute bidding" has been documented on Internet auctions by Bajari and Hortacsu (2003) and Roth and Ockenfels (2002). However, Internet auctions are open- ascending auctions where bidders can see and respond to each others bids, whereas Bank of Canada's auctions are sealed bid auctions in which dealers do not observe other dealers' bidding activity. Hence, "hiding information" from rivals, which is the primary explanation proposed by Bajari and Hortacsu (2003), should not be the primary concern of the securities dealers.

One explanation for the last-minute concentration of bids is that, especially for competitive bids, new information is very important in forming expectations about the appropriate value of the security being auctioned. Hence, bidders are reluctant to commit to a price until they are certain that no new information will be released.

An observation that appears consistent with this explanation is the difference between the bid submission times for long-term vs. short-term securities seen in Figure 1. The submission timings for long-term securities are much more concentrated at the very last minutes of the auction, with the median official bid for securities with maturity greater than one year coming 2.5 minutes before the auction deadline, as opposed to 9.3 minutes for securities with maturities less than or equal to one year. Pricing longer maturity securities depends quite sensitively on expectations about the future, and since many more factors enter into forming expectations about the long-term rather than the short-term, one may expect bidding decisions to be more responsive to arrival of new information.

We now bolster the information gathering hypothesis by presenting evidence that later bids are more informative of the auction outcome. To do this, we first categorized official bids in every auction by their submission times by dividing time into 5 minute periods leading up to the deadline. Since bids come in the form of multiple price-quantity pairs, i.e. demand schedules, we calculated the quantity weighted average price by the formula:

$$p^{QW} = \frac{\sum_{i=1}^{K} p_i q_i}{\sum_{i=1}^{K} i = 1^K q_i}$$
(1)

We then calculate the "informativeness" of each bid by calculating the absolute difference between p^{QW} and the "cutoff" price of the auction, i.e. lowest price at which the securities were sold. Since Bank of Canada securities auctions follow the discriminatory (pay-your-bid) pricing format, a bidder bidding right at the "cut-off-price" will make the highest profit (assuming, of course, all bidders derive the same value from winning the auction).

Figure 2 plots, as an average over all auctions in our data set, both the number of bids received in each 5 minute time interval, and their "informativeness," as measured by the absolute difference between the bid price and the cut-off price. Observe first that, consistent with the discussion above, many more bids arrive late than early. Observe also that the average "absolute prediction error" of bids declines from 15 basis points at the 30 minute mark to 2.5 cents within the last 5 minutes. This suggests that the early bids, on average, reflect valuations that are quite far from the "average market sentiment" reflected in the cut-off price in the auction.

In figure 3, we see that this "information aggregation at the last-minute" phenomenon is much more pronounced for long-term securities than for short-term securities auctions. In this figure, we compare a normalized informative-ness measure across these two types of securities, where we compute the ratio: $\frac{|p_t^{QW} - p^{cutoff}|}{|p_{Smins}^{QW} - p^{cutoff}|}$, i.e. the ratio of the absolute prediction error of the bids that are submitted at time t, and the absolute prediction error of the bids that are submitted in the last 5 minutes of the auction. Observe that this ratio is constant around 1 for short-term securities auctions, suggesting that earlier bids are as good predictors of the auction outcome as later bids. In contrast, for longer term securities, bids that come in more than 30 minutes before the auction deadline have absolute prediction errors that are 10 times as large as bids that come in the last 10 minutes.

5 Dispersion in Dealers' Bidding Performance

One explanation for the temporal patterns described in the previous section is that bid submissions respond to *publicly observed* information flows. All dealers taking part in the Bank of Canada securities auctions have access to global and local financial news sources, and can monitor trading activity between dealers on the electronic interdealer trading platforms.¹⁴ Since dealers acting strategically to maximize profits from the auction will try to make use of all the information they can get, they will wait until the last possible moment to cumulate these pieces of public information in their bids.

If all bidders have access to the same publicly available information, we might expect them to submit very similar bids in the auction, or, at least, give similar bids on average (i.e. there might be random deviations in each bidder's strategy from auction to auction, but, on average, bidders will submit price bids that are similar). The numbers in Table 1 (predictiondispersion.pdf), however, suggest otherwise. In this table, we calculated the absolute value of the difference between each dealer's quantity-weighted price bids and the cutoff price of the auction. We then averaged these "prediction error" measures over our subsamples of auctions of securities with similar maturities. The results in Table 1 suggest substantial dispersion in the cutoff-price prediction performance of the 21 dealers in our data set. In particular, the standard deviation of the dealers' performance is almost as large as the mean absolute prediction error.

 $^{^{14}{\}rm The}$ electronic interdealer market accounts for 86% of the volume traded on the interdealer market.

What may explain this large dispersion across dealers' bidding performances? If all bidders have access to the same information sources, and all of them value the security in the same way, it is difficult to fathom that they would bid prices that are very different from each others. Hence, one explanation is that different bidders have access to different, private information sources. Another explanation is that some bidders are simply better at utilizing (possibly publicly available) information to make a forecast about the auction cutoff price. This differential in forecast ability may come from having better skilled traders, or having better organizational decision making capabilities. There may also be a fixed cost associated with utilizing econometric/statistical technology to make forecasts. Hence, this second explanation may suggest a connection between bidder performance and the size of the bidder's operation.

We now investigate the determinants of the dispersion in the prediction performance of the dealers. The first step of this exercise is summarized in the series of regressions reported in Table 2. In the first column, we sought the correlation of dealer prediction performance with the (log) average size of the quantity bids placed by the dealer *for its own account* across all the auctions in our data set. Controlling for categorical variables accounting for different maturity levels, we find that dealer size is actually positively correlated with the size of the average prediction errors. This result appears contradictory with a theory of bidder performance in which size would become positively correlated with bidding performance through forces of competitive selection.

However, note that the "size" variable we have selected are the dealer's own bids, submitted, presumably, for speculative purposes. Another, perhaps more important, source of revenue for a dealer is to intermediate transactions by its customers who do not have direct access to bidding. The advantage from intermediating customer bids is that a dealer has the opportunity to get a "second opinion" as to what the market's valuation of the security being sold. Hence, the more "second opinions" that a dealer gets, the better its forecast may become.

One could easily rationalize this intuition with a standard Bayesian updating model of expectation formation, in which the dealer has a prior belief about the unknown security value, v, where the prior has mean μ and variance σ_v^2 . Suppose the dealer also has a private signal about the security value, x, which is distributed about the true value of the security with mean zero error, i.e. $x = v + \varepsilon_x$, where ε is distributed with zero mean and variance $\sigma_{\varepsilon_x}^2$. If all of the aforementioned random variables are distributed normally, conditional on this private information, the dealer expects the value of the security to be:

$$E(v|x) = \frac{\tau_v}{\tau_v + \tau_{\varepsilon_x}} \mu + \frac{\tau_{\varepsilon_x}}{\tau_v + \tau_{\varepsilon_x}} x$$
(2)

where $\tau_v = \frac{1}{\sigma_v^2}$ and $\tau_{\varepsilon_x} = \frac{1}{\sigma_{\varepsilon_x}^2}$, are the precisions of the prior and private signals respectively.

Now, if the dealer receives another signal about the security's true value, in the form of a customer bid, for example, she will incorporate this information into her conditional expectation. Specifically, call this piece of information y, where $y = v + \varepsilon_y$, also distributed with mean zero noise about the true security value. Given y, the conditional expectation becomes:

$$E(v|x,y) = \frac{\tau_v}{\tau_v + \tau_{\varepsilon_x} + \tau_{\varepsilon_y}} \mu + \frac{\tau_{\varepsilon_x}}{\tau_v + \tau_{\varepsilon_x} + \tau_{\varepsilon_y}} x + \frac{\tau_{\varepsilon_y}}{\tau_v + \tau_{\varepsilon_x} + \tau_{\varepsilon_y}} y \qquad (3)$$

i.e. a simple linear combination of the different pieces of information that the bidder received, weighted by their relative precisions.

It is also easy to see that the variance of this conditional expectation as an estimate of true value of the security, v, is lower when the dealer observes the additional piece of information y, as opposed to when she does not. In particular, the variance of the estimator should decline with the number of independent additional signals that the dealer observes.

The second and third columns of Table 2 appear to provide support to this intuition. In the second regression to explain differences in cutoff price prediction performance, we add a variable for the log of the average size of the customer bid relayed by the dealer. As suggested by the discussion above, the size of customer bids is negatively correlated with the prediction error. The addition of this variable appears also takes away some of the significance of the positive "own size" effect.

In the third column, we also add the (average) number of unique customers that a dealer serves into the prediction performance regression. We find that the coefficient on the number of unique customers is negative and significant at the 5-percent level, with the estimate suggesting that an additional customer reduces mean absolute prediction error by 9 cents! Interestingly, the negative coefficient on customer order size loses its significance, and declines in value, suggesting that the number of independent opinions contained in the customer opinions may matter more in decreasing predictive performance than the size of the customer orders.

The final column of Table 2 investigates a comparative static suggested by the discrepancy in temporal bidding patterns across long-term and short-term securities that we found in the previous section. Recall that we argued that the temporal patterns suggested that "information gathering" played a more important role for longer maturity securities than for short-term securities. If this argument is correct, we should also expect the "predictive advantage" from having an additional customer's opinion to be higher in longer maturity securities auctions than in short maturity auctions. Indeed, in the regression we run in the fourth column of Table 2 finds that the coefficient on the number of unique customers served by a dealer, is significantly negative only when interacted with an indicator for long-maturity securities auctions. The coefficient estimate suggests that for securities with maturity above 1 year, an additional customer accounts for an improvement in prediction accuracy of up to 24 cents – a rather large number when compared to the performance differentials reported in Table 1.

A final note about the regression in Table 2 is the interpretation of the positive coefficient of the dealer's own "size" variable. A dealer's demand for a given security may possess a private value component, as well as a pure common value component, as modelled above. For example, as is the case in Canada, a primary dealer may have to fulfill a minimum bidding requirement to preserve its privileged position. Hence, the primary dealer may find it necessary to knowingly bid above the cutoff price to fulfill this requirement. This would, of course, degrade the dealer's prediction performance as measured here. Since primary dealers are also the largest dealers in the market, this behavioral bias may explain the persistent positive correlation between size and prediction performance.

6 Dealer Bidding Behavior and Customer Orders

We now present more direct evidence for the hypothesis that found substantial support in the analysis of the previous section: "Dealers bid differently because they have access to different information sources. Two important sources of private information are customer bids and the forward (short or long) position of a dealer."

We will present two sets of evidence. We will first contrast the submission timings of dealer bids with the submission timings of customer bids, and present evidence that dealer bids lag customer bids. We will then argue that dealers' price bids reflect the informational content of customer bids, by showing that innovations in dealer bids follow innovations in customer bids very closely. We will also show that dealer bids are affected by the dealer's net short or long position.

6.1 Do Dealer Bids Follow Customer Bids?

Table 3 displays evidence for the hypothesis that dealers will submit their own bids after seeing customers' bids. To construct this table, we looked at all instances in which a dealer submitted own her own behalf as well as for a number of customers. We then compared the submission time of the latest customer bid to the submission time of the dealer's own bid using a pair-wise t-test (i.e. the within dealer difference). The test, when conducted for the entire spectrum of maturities reveals that dealer bids lag the latest customer bid by 0.43 minutes, the difference being statistically significant at the 1.7% level. Furthermore, we find that dealer bids lag customer bids 55% of the time.

However, as the next column of Table 3 shows, the difference in the timing between dealer and customer bids do not appear to statistically different for short-term securities. Dealer bids lag customer bids 52% of the time – not visibly (or, as it appears, statistically) different from an even split.

In contrast, dealer bids appear to lag customer bids much more visibly in auctions for longer-term securities. Not only both customer and dealer appear to come much later for these auctions, as was found in Section 4.1, dealer bids, on average, are submitted 2.5 minutes later than the latest customer bid. The lag is statistically significant at the 1% level. Moreover, dealer bids lag customer bids 74.9% of the time in these auctions.

Hence, the evidence suggests that dealers react to the information contained in their customers' bids *when and where it matters*. In particular, the second and third columns of Table 3 confirm our findings so far regarding the difference in the importance of "customer information" across longer term vs. shorter term securities auctions. Granted, given that customer bids also come very near the bid submission deadline, one may suspect that dealers will have little time to perform elaborate calculations. However, our data suggests that, on average, a minute or two appears to suffice.

6.2 Do Dealer Bids Respond To the Informational Content of Customer Bids?

We now investigate whether dealers change their bids in response to the arrival of customer bids, and whether the direction and magnitude of this change is explained by the informational content of the customer order. To do this, we utilize the bid history aspect of our data set, which comprises of *all* bids submitted by the dealers, not just the official bids. This allows us to track the modifications that dealers make in their bids on Bank of Canada's computerized bidding system.

To calculate modifications in dealers' bids, we fix a time interval, starting T = 10 or T = 30 minutes prior to the bid submission deadline, until the submission deadline. We then calculate the "standing bid" of the dealer at time T, which is the dealer's most recent bid as of time T. However, the dealer can change this bid until the bid submission deadline. Thus, we find the dealer's "official" bid, i.e. the final bid by the dealer that is accepted by the Bank of Canada. We then calculate the difference between the (natural logs) of the dealer's official bid and her standing bid at T minutes prior to the deadline. Since bids comprise of multiple price-quantity points, we take the quantity-weighted average price as the bid.

We then perform the same calculation for the customers, taking the (log) difference between their official bid and their standing bids at T minutes prior to the bid submission deadline. Thus, the change in a dealer's information set between time T and the auction deadline includes these modifications in customer bids.

Another piece of information that the dealer obtains within this time period is through the over-the-counter customer-dealer market, where the dealer engages in forward short or long trades. Unfortunately, we do not have access to this trading information. However, since by law dealers have to report changes in their net long or short positions to the Bank of Canada along with any changes to their bids, we observe net impact of this trading activity through modifications to the dealers positions. Again, we code these modifications as the difference between a dealer's "standing" net position at T minutes prior to the deadline, and the net position reported along with the dealer's official bid.

Aside from these two sources of information, which are privately observed, dealers may also modify their bids within the last T of an auction due to news coming from public information sources. Moreover, customers may also be modifying their bids in response to the release of the same public information – hence an observed correlation between dealer bid modifications and customer bid modifications may be due to the fact that both parties are responding to the same information). Hence, we have to control for the presence of public information within the chosen time interval to identify a causal relationship between customer bid modifications.¹⁵

We do this by utilizing an auction-level fixed effect specification. This amounts to looking at the variation across dealer bid modifications in the last T minutes within a given auction, and how much of this variation is explained by customer bid modifications observed by each dealer in this time period. One weakness of the auction-level fixed effect specification is that it assumes that the linear relationship between dealer bid modifications and customer bid modifications (and net position changes) is the same across different auctions in our data. We relax this assumption somewhat by conducting our analysis on Treasury bills and bonds separately.

Table 4 reports the results of the auction-level fixed effect regressions. The dependent variable in this table is the modification in a dealer's bid during the last 10 or 30 minutes of an auction. Note that in both time intervals, modifications in customer bids are positively and significantly correlated with modifications in dealer bids. The point estimates imply that a 1% increase in customers' price bids within the last 10 minutes implies a 0.297% increase in the dealer's bid. Within the last 30 minutes, the effect is smaller – a 1%

 $^{^{15}}$ We should note that most Central Banks including Canada, set auction deadline timing so that it does not coincide with major news releases like monetary policy target or income and employment data. However, it is possible that other sources of financial news, such as developments in U.S. financial markets, may affect asset prices in Canada. One way to control for the presence of public information flows is to proxy for this using real-time trading prices for securities that might be close substitutes to the security being auctioned. We actually observe this additional piece of data. In Canada, CanPX a data service, consolidates trade and quote data submitted by Canada's fixed income electronic interdealer-brokers. This information is disseminated to both customers and dealers.¹⁶ Potentially, both customers and dealers could track the impact of any publicly observed information shock on prices right uptill the auction deadline, and consequently make modifications in the bids they submit in the auction. However, when we looked at CanPX data to extract proxies for public information flows. Even though we have documented that 99% of the bid modifications are made in the last 30 minutes before the auction deadline, for each auction in the sample, we looked at CanPX data for a two-week window prior to the auction coinciding with the time when the when-issued market commences. We observed **no** trade or quote activity for the yet-to-be-auctioned security in the entire period covered by our sample. This empirical fact was puzzling given that the basis for a when-issued market is price discovery. However, these results are consistent withe the findings of Drudi and Massa (2001) that dealers try to hide orderflow information from other market participants in the Italian market for government securities.

upward revision in customers' bids translates into a 0.094% upward revision by the dealer. However, the fact that the correlation between dealer bid revisions and customer bid revision within the last 30 minutes is smaller than in the last 10 minutes is consistent with the observation that few customers are active at the 30 minute mark – thus modifications in these bids do not affect the dealer's bid as much as modifications by customers who arrive later. In fact, if when we compared the number of "strategic" customers (defined as those who modified their bids at least once within an auction) who were active (i.e. had submitted a bid) at T = 10 vs. T = 30, we found that more such customers were active at T = 10 as opposed to at T = 30.

When we conduct the auction-level fixed effects regression for the subsamples of Treasury bills and bonds separately, an interesting pattern emerges that appears consistent with our previous findings. We find that for both the T = 10 and T = 30 time intervals, the correlation between dealer bid modifications and customer bid modifications is much stronger in bond auctions than in Treasury bill auctions – in fact, the correlation in Treasury bill auctions is very near zero in both time intervals (although the correlation in the T = 30 interval for bonds was estimated with low precision owing to the small sample size, the point estimate is of the right sign). This is consistent with our earlier results suggesting that information contained in customer bids matters much more in bond auctions (which are harder to price since bond prices are, presumably, subject to more uncertainty) than in Treasury bill auctions.

Our regressions also indicate that changes in net positions of a dealer are important drivers of modifications in the dealer's bid. Once again, the effect appears much stronger in bond auctions as opposed to Treasury bill auctions – this is most likely due to the fact net positions, and thus net position changes in bond auctions are much larger (about 20 times) than in Treasury bill auctions (which is also driven by the fact that bond auctions are typically much larger on average than bill auctions). As for the sign of the estimated coefficient it is interesting to note the sign change across the last 10 minutes vs. the last 30 minutes in the (bond) auctions. Intuitively, it is not entirely clear what sign we should expect: assuming that the auction price is positively correlated with the secondary market trading price of the security, one might expect a dealer with a large short position to try to bid low so as to depress prices in the resale market, so that she can cover her short positions at as low a price as possible. On the other hand, one might expect a dealer with a large short position to bid high in the auction to cover as much of her short position to avoid being squeezed in the secondary market. Based on our results, unfortunately, we can not tell between these two possible hypotheses.

Taken together, the results of this section and section 5 suggest that customer bids provide private information to dealers and that dealers make use of their ability to observe customer bids. We also note that interactions with customers through forward (when-issued) transactions also drive dealer bids in this market.

7 Customers Respond to Dealer's Use of Order-Flow Information

In the previous sections, by looking at the dealer-side of the primary market, we found considerable support for the hypothesis that an important source of private information for dealers is customer bids, and that dealers use customer bids to revise their opinion about where the cutoff price in the auction might be. In this section we present evidence to support this hypothesis from the customer side of the primary market. Customers realize that dealers through whom they submit bids, revise their opinion about where the cutoff price will be, on the basis of customer bids. We present three features of customer bidding behaviour, which can be viewed as strategic responses of customers to bid revision by dealers on observing customer bids.

In section 7.1, we investigate the hypothesis that by holding-off bid submission till just before the auction deadline, customers can try to prevent dealers from making strategic use of their order flow information. Since a dealer wants to condition on as much customer information as possible, customers may try to push a dealer "beyond the deadline" by submitting their bids at the last minute, and force the dealer to risk giving his own bid later than the deadline. In Section 7.2, customers attempt to conceal their entire demand schedule from a single dealer by using multiple dealers to submit bids. In Section 7.3 we show that some customers are in a long-term relationship with a dealer in that they use a distinct dealer to submit their bids across all auctions. Presumably these repeated interactions provide dealers with valuable information about the cutoff price in the auction, for which they are likely to compensate the customers. We find that the payoff to the customer takes the form of the customer paying a lower price for the securities at the auction when he submits bids through his long-term dealer compared with other dealers, reflecting that the tendency to revise bids by dealers is less pronounced when the dealer's bid revision is conditional on the bid of a customer with whom he is in a long-term relationship. Finally, in parallel with the difference in the "information gathering" hypothesis between bonds and treasury bills, formulated in Section 5, we find that the strategic responses of the customer to bid revision by the dealer are much more pronounced in bond auctions compared with treasury bill auctions.

Thus, the customer-side of the primary market reflects the "information gathering" hypothesis formulated from the dealer-side in sections 5-6.

7.1 Last Minute Bidding by Customers Leads to Late Bids by Dealers

In Section 6.1 we observe that while customer bids lag dealer bids, customer bids come quite close to the submission deadline as well. A possible reason for this could be that customers attempt to conceal their demand curves from dealers as long as they can, as they realize that dealers will use these to revise

their own bids. We now look at late tenders to investigate this proposition: do customers bid close to the submission deadline to prevent dealers from revising their bids on seeing customer bids. Late tenders will refer to bids that arrived after the official bid,¹⁷ and were rejected by the issuer for being submitted after the submission deadline.

If customers could indeed prevent dealers from using their orderflow information by holding-off submission of their bids till just before the submission deadline, we would expect to see a significant number of late bids from dealers: dealers bids that incorporate the last-minute bids of customers arrive after the submission deadline, and are rejected. We actually find evidence to the contrary. Only 1% of the total tenders, and 2% of the official tenders submitted by the dealers were late. However, the fact that we see so few late tenders does not necessarily contradict our hypothesis that customers bid close to the auction deadline to conceal their demand curve from dealers. This is because a customer also knows that a bid that she submits too close to the submission deadline has a higher likelihood of not being transmitted by the electronic auction system, as the dealer through whom this bid has been submitted has hit his submission limit, the maximum amount he can bid as a submitter.¹⁸ In this instance bidding too close to the submission deadline would mean that the customer does not participate in the auction, and will have to buy the security at a higher price in the secondary market that follows the auction. If she has a large order, which is when she will attempt to bid as late as possible, the consequence of not participating can be particularly severe as this customer could be potentially "squeezed" in the secondary market.

Evidence that is more in line with the hypothesis that customers bid close to the auction deadline to prevent dealers from revising their bids emerges, when we look at the bidders from whom the late bids originate, and the auctions where the late bids are submitted. In case customers bid close to the bid submission deadline to prevent dealers from revising their bids on seeing customer bids, we should find that most of the late bids are dealer bids. We find that this is the case. 77% of the late tenders were submitted by the primary dealers and 13% by the customers.¹⁹

Finally, late bidding should be more pronounced for bond auctions rather than treasury bill auctions as there is greater uncertainty in pricing the former, and hence the tendency by customers to conceal their bids till the end should be more pronounced for bond auctions than treasury bill auctions. This is supported by the data as well. We find that the proportion of late tenders in

 $^{^{17}{\}rm The}$ official bid of a participant refers to the tender of this participant that is used to determine the cutoff yield.

¹⁸When a dealer submits a bid on behalf of a customer on the electronic auction system, and including this bid the dealer has bid in excess of the amount he is permitted as a submitter, the auction system automatically does not transmit this bid and flashes this message on the dealer's screen. Since these bids are not entered in the auction system, they are not recorded, and we could not analyze them.

 $^{^{19}{\}rm The}$ difference is statistically significant at the 99% level with the test statistic, Z=7.9.

bond auctions is more than double that in treasury bill auctions.²⁰ In addition late tenders were submitted in twice as many bond auctions compared with treasury bill auctions.²¹ If the value of orderflow information is greater in bond compared with treasury bill auctions as we have hypothesized, then we should find that the difference in the amount paid by the bidders when the status of the late tenders is changed from rejected to accepted is significantly greater for bond rather than treasury bill. We find this to be the case with the former being thrice the latter.²²

Thus, although there is some suggestive evidence that customers may be bidding close to the submission deadline to prevent dealers from revising their bids upon observing customer bids, though the evidence is far from being conclusive.

7.2 Multiple-Submitter Customers

In Section 7.1 we observe one aspect of strategic bidding by customers to conceal their demand curves from dealers: customers wait as late as they can to submit bids. In this section, we establish a second dimension of strategic bidding by customers: customers submit bids through more than one dealer to avoid revealing their demand curve to a single dealer.

There are two types of customers in our sample in terms of the average number of dealers through whom a customer submits bids in an auction. Table 5 plots the frequency distribution of customers in terms of average number of submitters used in an auction. It shows that about 10% of the customers used more than one submitter in an auction. Henceforth, customers who have a submitter average in an auction greater than or equal to 1.5, will be referred to as multiple-submitter customers, and those who have a submitter average of less than 1.5, will be referred to as single-submitter customers. This section establishes that multiple-submitter customers submit bids through more than one dealer to avoid revealing their demand curve to a single dealer as they realize that dealers use customer bids to form an opinion about the cutoff price at the auction.

Customers appear to use a larger number of dealers to submit bids when they have a larger quantity of the security to buy. Let S_t^i indicate the number of submitters used by customer *i* in auction *t*, q_t^i the total quantity demanded by customer *i* through all submitter used by the customer in auction *t*, and IA_t is the amount of the security issued in auction *t*. We also define MPS as the marginal propensity to submit,

$$MPS = \frac{\Delta S_t^i}{\Delta \frac{q_t^i}{IA_t}}$$

 $^{^{20}}$ For bonds, late tenders as a proportion of the number of auctions in the sample was 0.64; for treasury bills the corresponding proportion was 0.31. The difference is significant at 99% with the test statistic, Z=4.3.

 $^{^{21}}$ Late tenders were submitted in 50% of the bond auctions in the sample, and only 25% of the treasury bill auctions. The difference was significant at 99% with the test statistic, Z=3.4. 22 The difference is significant at the 95% level.

Column (1) of Table 7 reports the regression of the number of submitters, S_t^i , on the proportion of bid amount to issue amount, $\frac{q_t^i}{IA_t}$; the slope coefficient in this regression is the marginal propensity to submit. Customers use an additional dealer to submit tenders for a 36% increase in the ratio of bid amount to issue amount, with this coefficient being significant.

In general, a customer could be using multiple dealers as submitters in an auction due to either one or both of two effects: "concealing effect" or the "submission limit" effect. "Concealing effect" refers to a customer using multiple submitters in an auction to conceal his full demand curve from a single submitter. "Submission limit" effect refers to a customer using multiple submitters in an auction due to a binding constraint on a dealer as a submitter described in Section 2.

A first test, which indirectly supports the hypothesis that multiple-dealer customers submit bids through more than one dealer predominantly due to the "concealing effect", is based on a comparative static that exploits the differences in the temporal bidding patterns established for the dealer-side of the market in the previous sections. Our results suggested that "information gathering" by dealers from customers through which dealers obtain "second" opinion about the cutoff price price, played a more important role for bond auctions compared with treasury bill auctions. If these results were supported by the customerside of the market, the tendency to use a larger number of dealers in response to quantity demanded due to the "concealing effect" should be much more pronounced for bonds than treasury bills. That is, the marginal propensity to randomize should be greater for bonds than for bills for all customers. We re-run the regression in Column (1), Table 7 by interacting the ratio of bid amount to issue amount with an indicator for bond auctions. Results in Column (2), Table 7 show that this is the case. For bond auctions, customers use an additional dealer to submit tenders for a smaller increase in the ratio of bid amount to issue amount compared with treasury bill auctions, and the difference is statistically $significant.^{23}$ But this result could also suggest that customers demand is higher in bonds relative to treasury bills, and therefore it is the "submission limit" effect that leads them to submit bids through multiple dealers. We find the reverse: the ratio of bid amount to issue amount is 10% higher for treasury bills than bonds, and this difference is statistically significant.

A second test of the hypothesis that multiple-dealer customers submit bids through more than one dealer predominantly due to the "concealing effect", is based on a comparative static that exploits the difference in the marginal propensity to submit of single-submitter and multiple-submitter customers, in auctions where they submit bids through two or more dealers. In these auctions, both multiple-submitter customers and single-submitter customers use more than one submitter; but the former do it due to both the "concealing effect" and

 $^{^{23}}$ For bond auctions, customers use an additional dealer to submit tenders for a 24% increase in the ratio of bid amount to issue amount, and for treasury bill auctions they use an additional dealer for a 40% increase in the ratio of the bid amount to issue amount.

"submission limit" effect, and the latter do so only due to the "submission limit" effect. Thus, for each additional dollar of the security demanded, a multiplesubmitter customer will spread his bids across a larger number of submitters than a single-submitter customer reflecting that multiple-submitter customers use more than one submitter due to both the "concealing effect" and "submission limit" effect, whereas single-submitter customers use more than one submitter only due to the former. In short, a larger MPS of multiple-submitter customers relative to single-submitter customers in auctions where both use two or more dealers as submitters, supports the hypothesis that multiple-dealer customers submit bids through more than one dealer predominantly due to the "concealing effect".

Before testing this hypothesis, we need to control for customers who participate only in a few auctions in the sample. Customers who participate in less than 20% of the auctions of a specific security will be referred to as passive customers, and those who participate in more than 20% of the auctions of a specific security will be referred to as active customers.²⁴ A test based on the comparison of the MPS of single-submitter customers with that of multiple-submitter customers, but without controlling for the presence of passive customers, may not be conclusive. This follows from the proportion of passive customers to total customers being significantly greater for single-submitter than multiplesubmitter customers as documented in Table 6, and passive customers bidding differently from active customers. Since they participate in so few auctions, it is conceivable that passive customers do not learn that dealers are likely to use their orderflow information to revise their opinion about the cutoff yield; hence they do not use multiple submitters. In addition, we also find that passive customers demand smaller quantities than active customers on an average.²⁵

Controlling for passive customers, the regression in Column (1) of Table 6 is re-run to test for "concealing effect" and "submission limit" effect hypothesis. The sample consists of official tenders of customers in auctions where they use two or more dealers as submitters. Column (3) reports the results of this regression, with rows 2-5 reporting the MPS of multiple-submitter active, single-submitter active, multiple-submitter passive and single-submitter passive customers, respectively. While multiple-submitter active customers use an additional submitter for a 45% increase in the proportion of bid amount to issue amount, single-submitter active customers actually decrease the number of submitters with an increase in the proportion of bid amount to issue amount, with the difference in the MPS being significant.

 $^{^{24}}$ Auctions for 30 year GOC securities are the least frequent in our sample, with a total of 10 auctions. We need a customer's participation in at least 2 auctions to study strategic bidding across auctions. Thus 20% was selected as the cutoff participation level.

²⁵A test of the difference in the average of $\frac{q_t^i}{IA_t}$ between active and passive customers yields a test statistic of 8.79, with Prob(Z > 8.79) = 0.

7.3 Customers in Long Term Dealer Relationship

If customers use multiple dealers to conceal their demand schedules from customers, how do we explain the existence of active single-submitter customers? We find that some of the active single-submitter customers are in long-term relationship with their dealers, and are compensated by the dealers for the information provided by these repeated interactions. The payoff to these customers is primarily in the form of these customers paying lower prices at an auctions when they submit bids through the dealer with whom they are in longterm relationship, compared with when they submit bids in the same auction through other dealers. Thus, the bid revision by dealers on seeing customer bids is much more pronounced when customers in long-term relationship submit bids through dealers other than the long-term dealer. In keeping with the difference in "information aggregation" between bonds and treasury bills, we also find that the payoff hypothesis is confined to the bond auctions, with customers who are in long-term relationship in the treasury bill sector being compensated by the dealer in the bond sector.

To show the existence of a long-term relationship, we want to show that single-submitter customers who are active submit a large proportion of their bids through a distinct submitter throughout the sample.²⁶ Thus we are going to concentrate on customers in row 1, Table 5; these customers are single-submitter customers. Single-submitter active customers can be obtained from row 2, Table 6, with the "average" column indicating that 19% of the nonrandomizing customers are active on an average across maturity range. However Tables 5 and 6 do not indicate that when on an average one submitter was used, was that submitter a distinct submitter or not, which is what we need to establish the existence of a long-term relationship. Hence, for each of the single-submitter active customers, we construct the ratio of number of auctions in which a specific submitter was used, to the total number of auctions in which the customer participated, for each **distinct** submitter used by a single-submitter active customer. This is referred to as a customer's *submitter proportion*. For each customer, we tested if the highest and second-highest submitter proportion is significantly different. Customers for whom this is the case are customers with long-term relationship with a specific dealer in that they submit most of their tenders in the sample period through a distinct dealer. Table 8 lists these customers. For example, customer TEA is a single-submitter active customer who has a long-term relationship with dealer WXZ in treasury-bill auctions. TEA participates in 25% of the treasury bill auctions held in the sample period. In 78% of these treasury-bill auctions in which customer TEA participates, he uses dealer WXZ as the submitter.

Having established the existence of a long-term dealer relationship, we now explore what kind of payoffs a customer gets from being in this long-term relationship with a dealer. We test three hypothesis about whether the average price

 $^{^{26}}$ Passive single-submitter customers are excluded from this exercise as we cannot establish long-run relationship for these customers in our sample period.

paid by a customer differs depending on whether a customer is in a long-term dealer relationship or not.

The **first** hypothesis we test is whether in the sector where there is a longterm relationship, a customer pays a lower price when he submits tenders through the dealer with whom he is in a long-term relationship, compared with the dealers with whom he is not in a long term relationship. Table 9, columns (1) and (2) report the results of these tests. The sample for testing this hypothesis comprises of allotted tenders of only those customers who are in long term relationship, in the sector(s) where there is long term relationship. These tenders could be submitted either through the dealer with whom the customer is in a long term relationship, or through the dealer with whom the customer is not in a long term relationship. For each tender allotted to a customer in an auction,²⁷ we construct the quantity weighted average price; the difference in quantity weighted average price and the cutoff price of the auction in which this tender was allotted, referred to as mark-up, is our dependent variable. For the regression in Column (1), Table 9, the explanatory variable is a dummy variable ltC_ltD . This variable takes the value 1 if the tender allotted to the customer was submitted through the dealer with whom he is in a long-term relationship, and 0 if it is submitted through any other dealer. While customers in a long-term relationship pay 0.5 cents less when they submit tenders through their long-term dealer compared with other dealers, this difference is not signif $icant.^{28}$

But we have documented the difference in behaviour across maturity range in the analysis in the previous sections. Exploiting this difference, we re-run the regression in column (1), Table 9 introducing dummy variables for whether the long-term relationship is in the bond or the treasury bill sector. The results of this regression are reported in column (2), Table 9. BD_ltC_ltD (TB_ltC_ltD) is a dummy variable that takes value 1 if a customer in a long-term relationship in the bond (treasury bill) sector, submits the tender through the dealer with whom he is in long-term relationship in that sector, and takes value 0 if the tender is submitted through a dealer other than the long-term dealer. Similary, TB_ltC_nltD is a dummy variable that takes value 1 if a customer in a longterm relationship in the treasury bill sector, submits the tender through the dealer with whom he is not in a long-term relationship in that sector, and 0 if the tender is submitted through the dealer with whom he is in a long-term relationship.²⁹

In keeping with the spirit of the results in the previous sections, we now find differences emerging in the pay-off hypothesis, depending on whether the long-

 $^{^{27}{\}rm A}$ customer could be allotted more than one tender in an auction as he has the option of submitting his tenders through more than one dealer.

 $^{^{28}}t=-1.21$, Pr(t > 1.21) = 0.89.

 $^{^{29}}$ The constant in the regression in column (2) is the difference in the quantity weighted average price and cutoff price of allotted tenders of customers in long-term relationship in the bond sector, submitted through the dealers other than the dealer with whom they are in a long-term relationship in the bond sector.

term relationship is in the bond or treasury bill sector. We find that customers in long-term relationship in the bond sector, pay 3 cents less in bond auctions when they submit tenders through the dealer with whom they are in a long-term relationship compared with submitting tenders through other dealers, and this difference is significant. Customers in a long-term relationship in the treasury bill sector, pay 0.2 cents less in treasury bill auctions when they submit tenders through the dealer with whom they are in a long-term relationship compared with submitting tenders through other dealers; but this difference is not statistically significant.³⁰ These results hold when we control for differences in the quantity weighted average price and cutoff price between customers, irrespective of whether they submit tenders through long-term dealers or other dealers.³¹

From the results in columns (1)-(2), Table 9, we see that customers in a long-term relationship in the bond sector are compensated by the dealer with whom they are in long-term relationship through a lower price compared with when these customers submit tenders through other dealers. This result does not go through for treasury bills. This leads to the question whether customers in a long-term relationship in the treasury bill sector are being compensated in the bond sector by the dealer with whom they are in a long-term relationship. This is the **second** hypothesis we test, with the results for this hypothesis test reported in Table 9, column (5). The sample to test this hypothesis is built as follows. We first isolate customers in a long-term relationship in the treasury bill sector who participate in bond auctions but are not in a long-term relationship in the bond sector. Our sample comprises of allotted tenders of these customers in bond auctions. These tenders could be submitted through the dealer with whom this customer is in a long-term relationship in the treasury bill sector, or through other dealers. A dummy variable, $ltD_{-}tb$, which takes the value 1 for the former tenders and 0 for the latter tenders, is our explanatory variable.³² Our dependent variable is the quantity weighted average price minus the cutoff price of the tenders allotted to the customers in our sample. Column (5) reports the results of this regression. Even though the customer is in a long-term relationship in the treasury bill sector, this customer pays 9 cents less in bond auctions when he submits tenders in bond auctions through the

 $^{{}^{30}\}text{F}=0.21, F(1, 196, 0.95) = 3.84.$

³¹This is done by re-running the regression in column(2), Table 9 after adding a dummy variable for each of the customers listed in Table 8 to the set of explanatory variables in column (2), Table 9. Customers in long-term relationship in the bond sector still pay 3 cents less in bond auctions when they submit tenders through the dealer with whom they are in a long-term relationship compared with submitting tenders through other dealers, and this difference is significant. Customers in a long-term relationship in the treasury bill sector, pay 0.07 cents less in treasury bill auctions when they submit tenders through the dealer with whom they are in a long-term relationship compared with submitting tenders through the dealer with whom they are in a long-term relationship compared with submitting tenders through the dealer with whom they are in a long-term relationship compared with submitting tenders through the dealer with whom they are in a long-term relationship compared with submitting tenders through the dealer with whom they are in a long-term relationship compared with submitting tenders through the dealer with whom they are in a long-term relationship compared with submitting tenders through other dealers, and the difference is not significant. In both cases the significance of the difference, or the lack of it, changes dramatically compared to the results in column (2), Table 9 with t=-12.99 for bonds and $F \approx 0$ for treasury bills.

 $^{^{32}}$ Thus, the dummy variable takes the value 1 if the customer is submitting the tender in bond auctions through the dealer with whom he is in a long-term relationship in the treasury bill sector. It takes the value 0 if the tender is being submitted in a bond auction through any dealer other than the long-term dealer.

dealer with whom he is in a long-term relationship in the treasury bill sector. In addition, we find that the price paid by these customers in bond auctions is comparable to the price paid by customers in long-term relationship in bond auctions, when they submit tenders through dealers with whom they are in a long-term relationship in the bond sector.³³

The **third** pay-off hypothesis we test is whether a customer in a long-term dealer relationship pays a lower price compared with customers who are not in long-term dealer relationship. Table 9, columns (3) and (4), report the results of this test. The sample to test this hypothesis is built as follows. From the previous sample, we remove all tenders submitted by the customer through the dealer with whom he is not in a long-term relationship. This gives us allotted tenders of customers who are in long-term relationship, submitted through the dealer with whom they are in long-term relationship, in the sector in which the customer is in a long-term relationship. We add allotted tenders submitted by customers who are not in a long-term relationship in treasury bill and bond auctions. These customers either use more than one submitter in an auction (multiple-submitter customers with a submitter proportion greater than 1), or they use one submitter in an auction, but this submitter is not a **distinct** submitter.

Column (3) reports the results of the regression of the difference of quantity weighted average price and the cutoff price on the dummy variable ltC_{ltD} . This dummy variable takes the value 1 if the tender allotted to the customer was submitted through the dealer with whom he is in a long-term relationship, and 0 if the customer is not in a long-term relationship.³⁴ In the sector where there is a long-term relationship, customers in long-term relationship when submitting tenders through the dealer with whom they are in long-term relationship, pay 0.5 cents more than customers who are not in long-term dealer relationship, and this difference is significant at the 99% level.³⁵ In Column (4), Table 9 we run the same regression, but now introducing the distinction between treasury bills and bond auctions. In both the treasury bill and bond sector, customers in long-term relationship when they submit tenders through the dealer with whom they are in long-term relationship. For bonds (treasury bills), the former pay 1 (0.5) cents more than the latter, and this difference is statistically significant.

This result appears counterintuitive at first in that customers in long-term relationship pay a higher price for Government of Canada securities when submitting tenders through the long-term dealer, compared with customers who

³³While the former pay 4 cents above cutoff price, the latter pay 6 cents (this is obtained as the sum of the constant and the coefficient of the dummy variable bd_ltC_ltD in column (2), Table 9). In addition, the difference is not statistically significant at the 95% level; t=0.63, t(14, 0.95)=1.8, with the t-statistic calculated using pooled variance of the two sample of customers.

 $^{^{34}}$ The constant in the regression in column (3) is the difference in the quantity weighted average price and cutoff price of the customers who are not in a long-term relationship.

 $^{^{35}}t = 2.72, P(t > 2.72) = 0.997.$

are not in a long-term relationship. What might explain this result? Given the fact that the long-term customer and his dealer interact with each other in sectors other than Government of Canada securities, it is conceivable that the payoffs to the long-term customers are being given in these sectors by the long-term dealer. The second hypothesis tested above suggests that this could be the case. Alternatively, it is conceivable that a customer enters a long-term relationship with a dealer because institutional reasons specific to a customer's business push him to the inelastic part of his demand curve relative to customers who are not in a long-term relationship.³⁶

Thus, customers who are in a long-term relationship with a dealer, are adequately compensated by the dealer for the information provided by the customers in repeated interactions.

8 Conclusion

This paper establishes that in the issuance process of Government of Canada securities where direct access to primary issuance is restricted to authorized security dealers, "order-flow" information is potentially the key source of private information for these security dealers. "Order-flow" information is revealed to a security dealer through his interactions with customers, who can place bids in the auctions only through the authorized security dealer. Since each dealer interacts with a different set of customers, they, in effect, see different portions of the market demand and supply curves, leading to differing "private" inferences of where the equilibrium price might lie.

What if this source of private information was "dissipated" through a change in the mechanism to issue government securities? This could happen either by allowing direct access to customers to place bids in the primary issuance, and/or imposing some form of transparency obligations on the authorized security dealers with respect to their secondary market activity. For example, in Italy, the secondary market is transparent in that it is a centralized, regulated electronic screen-based market. However, only authorized dealers are allowed to place bids in primary issuance, and as we point out in the Introduction, Drudi and Massa (2001) show how authorized dealers use the discrepancy in transparency to obtain government securities in the less transparent primary market at below-market prices. In the U.S., customers are allowed to place bids in the primary issuance, making the primary market transparent. But unlike Italy, the secondary market is largely over-the-counter, with the customer-dealer interaction protocol requiring the customer to even reveal his intention to buy or sell when he requests a quote! The Italian and the U.S. comparison, along with several other countries examined by Sareen (2003), highlight that a mechanism for issuing government securities appears to retain privacy of a security dealer's

 $^{^{36}}$ For example, a customer that is a pension fund could be legally required to hold a certain proportion of its portfolio in the form of Government of Canada securities. This may not be the case for another customer that is a hedge fund.

"order-flow" in *at least* one the markets in which the dealer is participating. Which one will be less costly for the issuer? Which one will increase participation in the primary issuance? These questions will be explored in future research.

9 References

- Binmore, Kenneth and Joseph Swierzbinski, "Treasury auctions: Uniform or discriminatory?" *Review of Economic Design*, 2000, vol. 5, issue 4, pages 387-410.
- Drudi, Francesco and Massimo Massa. "Asymmetric Information and Trading Strategies: Testing Behavior on the Primary and Secondary T-Bond Markets around Auction Days," working paper, 2001.
- Easley, David, Maureen O'Hara, and P. S. Srinivas. "Option Volume and Stock Prices: Evidence on Where Informed Traders Trade." *Journal of Finance*, Vol. 53, No. 2. (Apr., 1998), pp. 431-465.
- Evans, Martin D.D. and Richard K. Lyons, "Order Flow and Exchange Rate Dynamics," *Journal of Political Economy*, February 2002, 170-180.
- Fevrier, Philippe, Raphaele Preguet, and Michael Visser. "Econometrics of Share Auctions," working paper, 2002.
- Hansch, O., N. Naik and S. Viswanathan, 1998, Do Inventories Matter in Dealership Markets? Evidence From The London Stock Exchange, *Journal of Finance*, 53, 1623-1656.
- Gordy, Michael B. "Structural Analysis of Reservation Price Policy in a Treasury Bill Auction," working paper, September 1994.
- Hortacsu, A. "Mechanism Choice and Strategic Bidding in Divisible Good Auctions: An Empirical Analysis of the Turkish Treasury Auction Market," working paper, 2002.
- Ito, Takatoshi, Richard K. Lyons, and Michael T. Melvin. "Is There Private Information in the FX Market? The Tokyo Experiment." *Journal of Finance*, June 1998, 1111-1130.
- Lyons, Richard K., *The Microstructure Approach to Exchange Rates*, MIT Press, December 2001.
- Massa, Massimo and Andrei Simonov, "The Sneaky, the Sleepy and the Skeptic: a Behavioral Model of Market Making: Evidence of Strategic Market Making on the Treasury Bond Market," working paper, December 2001.
- Massa, Massimo and Andrei Simonov. "Reputation and dealers' trading," Journal of Financial Markets, vol. 6 (2003), pp. 99-141.

O'Hara, Maureen, Market Microstructure Theory, Blackwell Publishers, 1995.

- Nyborg, Kjell G., Kristian Rydqvist, and Suresh M. Sundaresan. "Bidder Behavior in Multi-Unit Auctions - Evidence from Swedish Treasury Auctions." *Journal of Political Economy*, Vol. 110, April 2002.
- Sareen, S., Cross-Country Comparison of Models for Issuance of Government Securities, Unpublished manuscript, Bank of Canada, 2003.
- Sareen, S., Commodity Bundling in Government Securities Auctions, Unpublished manuscript, Bank of Canada, 2002.
- Sareen, S., Auctioning Government of Canada Securities: Can we do better, Unpublished manuscript, Bank of Canada, 2001.
- Umlauf, Steven. "An Empirical Study of the Mexican Treasury Bill Auctions," Journal of Financial Economics, 1993, 33, 313-340.



Figure 1: CDF of Bid Submission Times



Figure 2: Last Minute Information Aggregation



Figure 3: Sensitivity of Last Minute Bidding to Term Type

Maturity	# Dealers	Mean (cents.) S	Std. Dev. (c	Min (cents)	Max (cents)
1 Month	14	0.38	0.33	0.11	1.11
3 months	17	0.55	0.42	0.18	1.63
1 year	18	1.59	1.22	0.61	5.3
2 years	21	3.72	3.18	0.42	14.72
3 years	21	4.58	4.78	0.96	18.05
5 years	21	8.71	6.04	2.15	24.3
6 years	21	10.37	10.71	1.41	43.18
10 years	21	20.01	20.74	4.66	76.01
30 years	21	45.56	32.49	15.47	146.96

Table1: Average Absolute Prediction Error

The "prediction error" is calculated, for each auction, as the absolute value of the difference between the dealer's quantity-weighted average price bid and the cutoff price of the auction. This is averaged, for each dealer, over all auctions within the maturity category listed in the table.

	Table 2: Prediction Error	Regression		
	(1)	(2)	(3)	(4)
	Abs. Prediction Error	Abs. Prediction Error	Abs. Prediction Error	Abs. Prediction Error
log(Dealer Size)	0.016	0.030	0.036	0.001
	(0.006)**	(0.017)*	(0.017)**	(0.038)
log(Customer Size)		-0.027	-0.006	-0.003
		(0.013)**	(0.016)	(0.024)
No. of customers			-0.095	-0.006
			(0.045)**	(0.057)
Bond*Dealer Size				0.046
				(0.042)
Bond*log(Customer Size)				0.005
				(0.032)
Bond*No. of Customers				-0.243
				(0.089)***
Constant	-0.310	-0.126	-0.549	-0.038
	(0.130)**	(0.335)	(0.384)	(0.733)
Observations	175	111	111	111
R-squared	0.52	0.58	0.60	0.64

Standard errors in parentheses, * significant at 10%; ** significant at 5%; *** significant at 1%

Dependent variable is the (averaged over auctions) absolute value of the difference between quantity-weighted price bids and the auction's cutoff price.

Dealer Size is measured as the dollar amount (averaged over auctions) that a dealer placed bids for on its own behalf.

Customer Size is measured as the dollar amount (averaged over auctions) that a dealer placed bids for on behalf of its customers.

Number of Customers is the average number of unique customers that a dealer serves per auction.

Bond is a dummy variable that equals 1 for securities that are more than 1 year in maturity

Maturity fixed effects are supressed

	All Maturities	Bills	Bonds
Submission Time of	8.64	9.59	2.27
Dealer Bid	s.d.= 7.92	s.d.= 8.03	s.d.= 2.15
(minutes from deadline)			
Submission Time of	9.08	9.71	4.81
Customer Bid	s.d.= 6.81	s.d.=6.87	s.d.= 4.53
(minutes from deadline)			
Difference	-0.43***	-0.12	-2.54***
Std.err.	0.20	0.23	0.29
P-value	0.02	0.3	0
% of times Dealer Bid	55.29%	52.39%	74.90%
precedes Customer Bid			
Number of comparisons	2042	1779	263

Table 3: Comparison of the Timing of Dealer vs. Customer

"Submission Time of Dealer Bids" measures the time, in minutes, before the auction deadline at which the last dealer bid is submitted. "Submission Time of Customer Bid" measures the time, in minutes, before the auction deadline at which the last customer bid is submitted, for the same auction and same dealer as above.

The reported "Difference" is the result of a pairwise t-test. Standard errors and p-value of the test are reported.

Table 4: What Drives Dealers' Bid Revisions?

Dependent variable: % Change in Dealer's Bid During Period

	Last 10 Minutes of the Auction			Las	Last 30 Minutes of the Auction		
	Entire Sample	T-Bills	Bonds	Entire Sample	T-Bills	Bonds	
% Change in Customers'	0.297***	0.0046	0.299***	0.094***	-0.011	0.054	
Bids During Period	(0.033)	(0.0074)	(0.098)	(0.016)	(0.050)	(0.049)	
(averaged over customers)							
Change in Dealer's	0.0177***	-1.81e-07	0.186***	-0.016***	-0.00037***	-0.015***	
Net Position	(0.0057)	(0.00006)	(0.054)	(0.001)	(0.00006)	(0.004)	
Number of Unique Customer	0.00003	-1.63e-06	0.00018	0.000059	2.73e-08	0.0004	
Bids Received by Dealer	(0.0001)	(1.12e-06)	(0.00089)	(0.00004)	(1.17e-06)	(0.0004)	
Number of Customers	0.00009	4.02e-06**	-0.00041	-0.00015	-1.94e-06	0.0016**	
Revising Bids During Period	(0.0002)	(1.59e-06)	(0.00110)	(0.000097)	(1.87e-06)	(0.0006)	
Observations	1918	1672	246	1909	1667	242	
No. of Auctions	213	153	60	213	153	60	
Avg. Observations/Auction	9	10.9	4.1	9	10.9	4	
R-sq: within	0.0508	0.0048	0.1044	0.1561	0.0201	0.2187	
between	0.0372	0.0004	0.0424	0.0370	0.0134	0.0150	
overall	0.0533	0.0031	0.1040	0.1427	0.0202	0.1853	

Notes: Each column reports the estimated coefficients from a fixed effects regression. The dependent variable is the percent change (calculated as a log first difference) between a dealer's standing price bid at 10 or 30 minutes before the bid submission deadline, and the dealer's official bid. The first independent variable is the percent change between this dealer's customers' price bids during the last 10 or 30 minutes in the auction. The second independent variable is the change in the dealer's reported net long (or short) position during the last 10 or 30 minutes of the auction. We also control for the number of unique customer bids received by the dealer, and the number of customers who revised their bids within the last 10 or 30 minutes in the auction. To purge the effect of public information sources in comovements across customer and dealer bid changes, we control for auction-level fixed effects in the regressions. Standard errors are reported below the coefficients. Significance at 10-percent, 5-percent and 1-percent leves are denoted by *,**, and ***.

Table 5: Frequency Distribution of Customers Who Participated inAt least One Auction in Terms of Average # of Submitters

			- //
Avg $\#$ of Submitters	Bonds	Treasury Bills	Customer Type
< 1.5	74	95	Single-Submitter
1.5 & < 2.5	18	18	Multiple-Submitter
2.5 & < 3.5	6	8	Multiple-Submitter

The numbers in this table referr to unique customer participation and not unique customers. Suppose a customer ABC participates in 30 year bond auctions, and 1 year treasury bill auctions in the sample. Then we obtain the average number of submitters used by this customer in the 30 year bond auction and the 1 year treasury bill auction. Suppose the average number of submitters used by this customer is 3 for the former and 1 for the latter auctions. This customer will then appear in row (1) under the "Treasury Bill" column, and row (3) undeer the "Bonds" column.

Table 6: Active vs Passive Customers

	Bond	Treasury Bill	Average
$\frac{\# \text{ of multiple-submitter passive customers}}{\# \text{ of multiple-submitter customers}}$	0.4	0.57	0.49
$\frac{\# \text{ of single-submitter passive customers}}{\# \text{ of single-submitter customers}}$	0.82	0.80	0.81

The column "average" is the average ratio across maturity range. The difference in the "average" ratio between multiple-submitter passive and single-submitter passive customers is significant at 1%, with Z = 34.01.

	(1)	(2)	(3)
bid amount	2.6**	2.53	
Issue Amount	(0.403)	(0.41)	
$\left(\frac{\text{bid amount}}{\text{Lague Amount}}\right)^*(\text{Bond})$. ,	1.69^{*}	
Allount X		(1.01)	
$\left(\frac{\text{bid amount}}{\text{Issue Amount}}\right)^*$ (Multiple-Submitter Active)		~ /	2.24^{**}
issue Amount			(0.8)
$\left(\frac{\text{bid amount}}{\text{Issue Amount}}\right)^*$ (Single-Submitter Active)			-3.49^{**}
			(1.04)
$\left(\frac{\text{bid amount}}{\text{Issue Amount}}\right)^*$ (Multiple-Submitter Passive)			-2.1
			(1.74)
$\left(\frac{\text{bid amount}}{\text{Issue Amount}}\right)^*$ (Single-Submitter Passive)			-8.01^{*}
			(3.68)
constant	1.329	1.32	2.6
	(0.029)	(0.03)	(0.06)
Observations	1413	1413	385
R-squared	0.0285	0.03	0.08
$F^*_{msa-ssa}$			24.45^{**}

 Table 7: Customers Use Multiple Dealers

** significant at 99%; * significant at 95%; standard error are given in parenthesis. Dependent variable is S_t^i , the number of submitters used by customer i in auction t. $\frac{\text{bid amount}}{\text{Issue Amount}}$ is the ratio $\frac{q_t^i}{IA_t}$, the amount bid by customer i in auction t over the amount issued in auction t. (Single-Submitter Active), (Single-Submitter Passive), (Multiple-Submitter Active), and (Multiple-Submitter Passive) are 0-1 dummy variables that take a value of 1 if in auction t, customer i is a single-submitter active customer, single-submitter passive customer, multiplesubmitter active customer, and multiple-submitter passive customer, respectively, and zero otherwise. The sample in columns (1)-(2) consists of official tenders of all customers. The sample in column (3) consists of official tenders of all customers in auctions where they submit tenders through two or more dealers. $F_{msa-ssa}^*$ is the F-statistic for the difference in the marginal propensity to submit, MPS, of multiple-submitter active and single-submitter active customers.

Customer	Maturity Range	Submitter	Submitter	Participation
			Proportion	Proportion
TEA	T-Bill	WXZ	0.782	0.245
OEO	T-Bill	DZZ	0.667	0.217
GUO	T-Bill	JDW	0.851	0.566
ZXN	T-Bill	KVY	0.778	0.248
TDI	Bond	DZZ	0.79	0.33
ZAW	Bond	DZZ	0.883	0.286
EQV	T-Bill	KVY	1.000	0.229

 Table 8: Customers with Long Term Dealer Relationship

	(1)	(2)	(3)	(4)	(5)
ltC_ltD	-0.005		0.005		
	(0.0041)		(0.0016)		
BD_ltC_ltD		-0.03		0.01	
		(0.012)		(0.006)	
TB_ltC_ltD		-0.086		-0.042	
		(0.010)		(0.003)	
TB_ltC_nltD		-0.085		× /	
		(0.011)			
$TB_{-}nltC$				-0.046	
				(0.0022)	
ltD th				(010022)	-0.09
					(0.059)
constant	0.0137	0.092	0.0067	0.05	0.125
	(0.0036)	(0.01)	(0.0007)	(0.002)	(0.029)
Observations	196	196	896	896	25
B-squared	0.008	0.44	0.008	0.35	0.09
$(\frac{\mu_0}{\mu_0}, \frac{\mu_0}{\mu_0}, \frac{\mu_0}{\mu_0}, \frac{\mu_0}{\mu_0})$	0.005	0.11	0.000	0.00	0.00
$(ltC_ltD - ltC_nltD)$	-0.005				
	(-1.21)	0.00***			
$(ltC_ltD - ltC_nltD)_BD$		-0.03			
		(2.55)			
$(ltC_ltD - ltC_nltD)_TB$		-0.002			
		(0.206)			
$(ltC_ltD - nltC)$			0.005***		
			(2.72)	0.044444	
$(ltC_ltD - nltC)_BD$				0.01***	
				(2.03)	
$(ltC_ltD - nltC)_TB$				0.005^{**}	
				(12.22)	
$(ltD_tb - nltD_tb)$					-0.09^{*}
					(-1.49)

Table 9: Payoff to Customers with Long Term Dealer Relationship

* * * significant at 99%; ** significant at 95%; * significant at 90%; standard error in parenthesis above the double lines. Dependent variable is the difference in the quantity weighted average price of the allotted tenders and the cutoff price in Canadian dollars. Both the quantity weighted average price and the cutoff price are quoted in terms of \$(CD) 100 of the security allotted.

Table 9 continued..

The sample for the regressions in columns (1) and (2) comprises of allotted tenders of only those customers who are in long term relationship, in the sector(s) where there is long term relationship. The sample for the regressions in columns (3) and (4) comprises of: allotted tenders of customers who are in long-term relationship, submitted through the dealer with whom they are in long-term relationship, in the sector in which the customer is in a long-term relationship; allotted tenders submitted by customers who are not in a long-term relationship in either the bond or the treasury bill sector. The sample for the regression in column (5) comprises of tenders allotted in bond auctions to customers who are in a long-term relationship in the treasury bill sector. ltC_ltD takes the value 1 if a customer in a long-term relationship submits the tender through the long-term dealer in the sector where there is a long-term relationship, and 0 otherwise. BD_ltC_ltD (TB_ltC_ltD) is the same dummy variable with the added condition that the long-term relationship is in the bond (treasury bill) sector. $TB_{-lt}C_{-nlt}D$ takes value 1 if the allotted tender of the customer in a long-term relationship in the treasury bill sector is submitted in treasury bill auctions through dealers other than the one with whom he is in a long-term relationship, and 0 otherwise. TB_nltC takes value 1 if the allotted tender is submitted by a customer who is not in a long-term relationship in the treasury bill sector, and 0 otherwise. ltD_tb takes value 1 if in bond auctions, the allotted tender of a customer in a long-term relationship in the treasury bill sector is submitted through the dealer with whom he is in a long-term relationship, and zero otherwise. Rows below the double lines compare the difference in the quantity weighted average price of the allotted tenders and the cutoff price, for two sets of customers. Row (1) is this difference for customers in a long-term relationship when they submit tenders through the long-term dealer, compared with when they submit tenders through other dealers, in the sector where there is long-term relationship. Rows (2) and (3) are identical to Row (1), except that they pertain to the long-term relationship being in the bond and the treasury bill sectors, respectively. Row (4) is the difference for customers in a long-term relationship when they submit tenders through the long-term dealer, compared with customers who are not in a long-term relationship, in auctions belonging to the sector where there is long-term relationship. Rows (5) and (6) are identical to Row (4), except that they pertain to the long-term relationship being in the bond and the treasury bill sectors, respectively. Finally, row (7) pertains to customers who are in a long-term relationship in the treasury bill sector; it is the difference in the quantity weighted average price and cutoff price of tenders submitted in a bond auction through the dealer with whom they are in a long-term relationship in the treasury bill sector, compared with any dealer other than the long-term dealer. Rows (1)-(3), rows (4)-(6), and row (7), test the first, third, and the second pay-off hypothesis described in Section 7.3. Test statistics for testing the significance of the difference across two sets of customers described above, are given in parenthesis.

NOTE DI LAVORO DELLA FONDAZIONE ENI ENRICO MATTEI

Fondazione Eni Enrico Mattei Working Paper Series

Our Note di Lavoro are available on the Internet at the following addresses:

http://www.feem.it/Feem/Pub/Publications/WPapers/default.html http://www.ssrn.com/link/feem.html

NOTE DI LAVORO PUBLISHED IN 2003

PRIV	1.2003	Gabriella CHIESA and Giovanna NICODANO: Privatization and Financial Market Development: Theoretical Issues
PRIV PRIV	2.2003 3.2003	<i>Ibolya SCHINDELE</i> : <u>Theory of Privatization in Eastern Europe</u> : <u>Literature Review</u> <i>Wietze LISE, Claudia KEMFERT and Richard S.J. TOL</i> : <u>Strategic Action in the Liberalised German Electricity</u>
CLIM	4.2003	<u>Market</u> Laura MARSILIANI and Thomas I. RENSTRÖM: Environmental Policy and Capital Movements: The Role of
		Government Commitment
KNOW	5.2003	Reyer GERLAGH: Induced Technological Change under Technological Competition
ETA	6.2003	Efrem CASTELNUOVO: Squeezing the Interest Rate Smoothing Weight with a Hybrid Expectations Model
SIEV	7.2003	Anna ALBERINI, Alberto LONGO, Stefania TONIN, Francesco TROMBETTA and Margherita TURVANI: The Role of Liability, Regulation and Economic Incentives in Brownfield Remediation and Redevelopment:
		Evidence from Surveys of Developers
NRM	8.2003	Elissaios PAPYRAKIS and Reyer GERLAGH: <u>Natural Resources: A Blessing or a Curse?</u>
CLIM	9.2003	A. CAPARROS, JC. PEREAU and T. TAZDAÏT: North-South Climate Change Negotiations: a Sequential Game
		with Asymmetric Information
KNOW	10.2003	<i>Giorgio BRUNELLO and Daniele CHECCHI</i> : <u>School Quality and Family Background in Italy</u>
CLIM	11.2003	Efrem CASTELNUOVO and Marzio GALEOTTI: Learning By Doing vs Learning By Researching in a Model of
		Climate Change Policy Analysis
KNOW	12.2003	Carole MAIGNAN, Gianmarco OTTAVIANO and Dino PINELLI (eds.): Economic Growth, Innovation, Cultural Diversity: What are we all talking about? A critical survey of the state-of-the-art
KNOW	13.2003	Carole MAIGNAN, Gianmarco OTTAVIANO, Dino PINELLI and Francesco RULLANI (lix): Bio-Ecological
KNOW	14.2003	Diversity vs. Socio-Economic Diversity. A Comparison of Existing Measures Maddy JANSSENS and Chris STEYAERT (lix): Theories of Diversity within Organisation Studies: Debates and
		Future Trajectories
KNOW	15.2003	<i>Tuzin BAYCAN LEVENT, Enno MASUREL and Peter NIJKAMP</i> (lix): <u>Diversity in Entrepreneurship: Ethnic and</u> <u>Female Roles in Urban Economic Life</u>
KNOW	16.2003	Alexandra BITUSIKOVA (lix): Post-Communist City on its Way from Grey to Colourful: The Case Study from Slovakia
KNOW	17.2003	Billy E. VAUGHN and Katarina MLEKOV (lix): A Stage Model of Developing an Inclusive Community
KNOW	18.2003	Selma van LONDEN and Arie de RUIJTER (lix): Managing Diversity in a Glocalizing World
Coalition		
Theory	19.2003	Sergio CURRARINI: On the Stability of Hierarchies in Games with Externalities
Network		
PRIV	20.2003	Giacomo CALZOLARI and Alessandro PAVAN (lx): Monopoly with Resale
PRIV	21.2003	Claudio MEZZETTI (lx): Auction Design with Interdependent Valuations: The Generalized Revelation
		Principle, Efficiency, Full Surplus Extraction and Information Acquisition
PRIV	22.2003	Marco LiCalzi and Alessandro PAVAN (lx): <u>Tilting the Supply Schedule to Enhance Competition in Uniform</u> - Price Auctions
PRIV	23.2003	David ETTINGER (lx): Bidding among Friends and Enemies
PRIV	24.2003	Hannu VARTIAINEN (lx): Auction Design without Commitment
PRIV	25.2003	Matti KELOHARJU, Kjell G. NYBORG and Kristian RYDQVIST (lx): Strategic Behavior and Underpricing in
		Uniform Price Auctions: Evidence from Finnish Treasury Auctions
PRIV	26.2003	Christine A. PARLOUR and Uday RAJAN (lx): Rationing in IPOs
PRIV	27.2003	Kjell G. NYBORG and Ilya A. STREBULAEV (lx): Multiple Unit Auctions and Short Squeezes
PRIV	28.2003	Anders LUNANDER and Jan-Eric NILSSON (IX): Taking the Lab to the Field: Experimental Tests of Alternative
		Mechanisms to Procure Multiple Contracts
PRIV	29.2003	TangaMcDANIEL and Karsten NEUHOFF (lx): Use of Long-term Auctions for Network Investment
PRIV	30.2003	Emiel MAASLAND and Sander ONDERSTAL (lx): Auctions with Financial Externalities
ETA	31.2003	Michael FINUS and Bianca RUNDSHAGEN: A Non-cooperative Foundation of Core-Stability in Positive
		Externality NTU-Coalition Games
KNOW	32.2003	Michele MORETTO: Competition and Irreversible Investments under Uncertainty
PRIV	33.2003	Philippe QUIRION: Relative Quotas: Correct Answer to Uncertainty or Case of Regulatory Capture?
KNOW	34.2003	<i>Giuseppe MEDA, Claudio PIGA and Donald SIEGEL</i> : <u>On the Relationship between R&D and Productivity: A</u> Treatment Effect Analysis
ETA	35.2003	Alessandra DEL BOCA, Marzio GALEOTTI and Paola ROTA: Non-convexities in the Adjustment of Different
		Capital Inputs: A Firm-level Investigation

GG	36.2003	Matthieu GLACHANT: Voluntary Agreements under Endogenous Legislative Threats
PRIV	37.2003	Narjess BOUBAKRI, Jean-Claude COSSET and Omrane GUEDHAMI: Postprivatization Corporate Governance:
~ ~ ~		the Role of Ownership Structure and Investor Protection
CLIM	38.2003	Rolf GOLOMBEK and Michael HOEL: Climate Policy under Technology Spillovers
KNOW	39.2003	Slim BEN YOUSSEF: Transboundary Pollution, R&D Spillovers and International Trade
CTN	40.2003	Carlo CARRARO and Carmen MARCHIORI: Endogenous Strategic Issue Linkage in International Negotiations
KNOW	41.2003	Sonia OREFFICE: Abortion and Female Power in the Household: Evidence from Labor Supply
KNOW	42.2003	Timo GOESCHL and Timothy SWANSON: On Biology and Technology: The Economics of Managing Biotechnologies
ETA	43.2003	Giorgio BUSETTI and Matteo MANERA: STAR-GARCH Models for Stock Market Interactions in the Pacific Basin Region, Japan and US
CLIM	44.2003	Katrin MILLOCK and Céline NAUGES: The French Tax on Air Pollution: Some Preliminary Results on its Effectiveness
PRIV	45.2003	Bernardo BORTOLOTTI and Paolo PINOTTI: The Political Economy of Privatization
SIEV	46.2003	Elbert DIJKGRAAF and Herman R.J. VOLLEBERGH: <u>Burn or Bury? A Social Cost Comparison of Final Waste</u> Disposal Methods
ETA	47.2003	Jens HORBACH: Employment and Innovations in the Environmental Sector: Determinants and Econometrical Results for Germany
CLIM	48.2003	Lori SNYDER, Nolan MILLER and Robert STAVINS: The Effects of Environmental Regulation on Technology Diffusion: The Case of Chlorine Manufacturing
CLIM	49.2003	Lori SNYDER, Robert STAVINS and Alexander F. WAGNER: Private Options to Use Public Goods. Exploiting Revealed Preferences to Estimate Environmental Benefits
CTN	50.2003	László Á. KÓCZY and Luc LAUWERS (Ixi): The Minimal Dominant Set is a Non-Empty Core-Extension
CTN	51.2003	Matthew O. JACKSON (1xi): Allocation Rules for Network Games
CTN	52.2003	Ana MAULEON and Vincent VANNETELBOSCH (1xi): Farsightedness and Cautiousness in Coalition Formation
CTN	53,2003	Fernando VEGA-REDONDO (1xi): Building Up Social Capital in a Changing World: a network approach
CTN	54.2003	Matthew HAAG and Roger LAGUNOFF (1xi): On the Size and Structure of Group Cooperation
CTN	55.2003	Taiji FURUSAWA and Hideo KONISHI (1xi): Free Trade Networks
CTN	56.2003	Halis Murat YILDIZ (1xi): National Versus International Mergers and Trade Liberalization
CTN	57.2003	Santiago RUBIO and Alistair ULPH (lxi): <u>An Infinite-Horizon Model of Dynamic Membership of International</u> Environmental Agreements
KNOW	58.2003	Carole MAIGNAN, Dino PINELLI and Gianmarco I.P. OTTAVIANO: <u>ICT, Clusters and Regional Cohesion: A</u> Summary of Theoretical and Empirical Research
KNOW	59 2003	Giorgio BELLETTINI and Gianmarco I.P. OTTAVIANO: Special Interests and Technological Change
FTA	60 2003	Romie SCHÖR: The Double Dividend Hynothesis of Environmental Taxes: A Survey
CLIM	61 2003	Michael FINUS, Feko van IERLAND and Robert DELLINK. Stability of Climate Coalitions in a Cartel
CLIM	01.2005	Formation Game
GG	62.2003	Michael FINUS and Bianca RUNDSHAGEN: How the Rules of Coalition Formation Affect Stability of
SIEV	63 2003	Alberto PETRUCCI: Taxing Land Rent in an Onen Economy
	64 2003	Joseph F AI DV Scott RAPPETT and Pohert N STAVINS: Thirteen Plus One: A Comparison of Global Climate
CLIW	04.2003	Policy Architectures
SIEV	65 2003	<i>Edi DEFERANCESCO</i> : The Beginning of Organic Fish Farming in Italy
SIEV	66 2003	<i>Ear DEF REPORTED</i> The Degramme of Organic Fish relation when Consumers Care for the Environment
SIEV	67 2003	Raus CONAD. The Competition and Flocket Differentiation when containers call of the Environment Paulo ALD NUMER Luca POSSETTO Avianno DE RUAELE Monetory Value Assessment of Clam Eiching
SILV	07.2003	Management Practices in the Venice Lagoon: Results from a Stated Choice Exercise
CLIM	68.2003	ZhongXiang ZHANG: Open Trade with the U.S. Without Compromising Canada's Ability to Comply with its Kyoto Target
KNOW	69.2003	David FRANTZ (lix): Lorenzo Market between Diversity and Mutation
KNOW	70.2003	Ercole SORI (lix): Mapping Diversity in Social History
KNOW	71.2003	Liiliana DERU SIMIC (1xii): What is Specific about Art/Cultural Projects?
KNOW	72.2003	Natalya V. TARANOVA (lxii): The Role of the City in Fostering Intergroup Communication in a Multicultural
KNOW	73.2003	Environment: Saint-Petersburg's Case Kristine CRANE (Ixii): The City as an Arena for the Expression of Multiple Identities in the Age of
KNOW	74 2002	<u>Viouansanon and Migration</u> Kazuma MATORA (Ixii): Glocal Dialogue- Transformation through Transcultural Communication
KNOW	75.2003	Catarina REIS OLIVEIRA (Ixii): Immigrants' Entrepreneurial Opportunities: The Case of the Chinese in
KNOW	76 2002	Portugal
KNUW	/0.2003	Sanara wALLMAN (IXII): The Diversity of Diversity - towards a typology of urban systems
KNUW	//.2003	KICHARA FEARCE (IXII): A BIOLOGIST S VIEW OF INDIVIDUAL CUltural Identity for the Study of Cities
KNOW	/8.2003	Vincent MERK (IXII): Communication Across Cultures: from Cultural Awareness to Reconciliation of the Dilemmas
KNOW	79.2003	Giorgio BELLETTINI, Carlotta BERTI CERONI and Gianmarco I.P.OTTAVIANO: Child Labor and Resistance to Change
ETA	80.2003	Michele MORETTO, Paolo M. PANTEGHINI and Carlo SCARPA: Investment Size and Firm's Value under Profit Sharing Regulation

	1000	Carlo CARRARO, Alessandro LANZA and Valeria PAPPONETTI: One Thousand Working Papers
CIN	125.2005	Players
CTN	122.2003	Edward CARTWRIGHT and Myrna WOODERS. On Equilibrium in Fure Strategies in Games with Many Players
CTN	122 2002	<u>Flayels</u> Edward CARTWRIGHT and Myrna WOODERS: On Equilibrium in Dura Stratagias in Camao with Many Disease
UIN	121.2003	Myrna WOODERS, Eawara CARIWRIGHT and Reinnard SELTEN: Social Conformity in Games with Many
CIN	120.2003	Gianiuigi VERNASCA: Dynamic Price Competition with Price Adjustment Costs and Product Differentiation
CTN	120 2002	Games with Incomplete Information: Some First Results
CIN	119.2003	<i>Eaward CARTWRIGHT and Myrna WOODERS</i> : <u>Social Conformity and Bounded Rationality in Arbitrary</u>
CIN	118.2003	Marita LAUKKANEN: Iransboundary Fisheries Management under Implementation Uncertainty
CIN	117.2003	Somaeb LAHIKI: Stable Matchings for a Generalized Marriage Problem
CTN	116.2003	Somdeb LAHIKI: Stable Matchings for the Room-Mates Problem
CIN	115.2003	Hideo KONISHI and M. Utku UNVER: Credible Group Stability in Multi-Partner Matching Problems
CIN	114.2003	Guillaume HAEKINGER and Myrna WOODERS: Decentralized Job Matching
CTN	114 2002	International Environmental Treaties
UIN	115.2003	Carlo CARRARO, Carmen MARCHIORI and Sonia OREFFICE: Endogenous Minimum Participation in
EIA CTN	112.2003	Jack GOODI (IXIV): GIODALISATION, POPULATION AND ECOLOgy Carlo CAPPAPO, Carmon MARCHIOPL and Sonia OPEEELCE: Endogonous Minimum Doministration in
	111.2003	Unaries FERRINGS and Brian WALKER (IXIV): Conservation and Optimal Use of Kangelands
NEM	110.2003	Anastasios AETATADEAS and Catarina KOSETA-TALMA(IXIV): <u>Instabilities and Kobust Control in Fisheries</u> Charles PEPPINGS and Brian WALKER (Ixiv): Conservation and Ontimel Use of Pengelande
NDM	110 2002	<u>Sustainable Development in Imperiect Economies</u>
SIEV	109.2003	Sustainable Development in Imperfect Economies
SIEV	100.2003	Kannath ARROW Partha DASCHPTA and Karl-Göran MÄLERAvivi): Evaluating Projects and Associate
SIEV	108 2003	Sara ANIYAR (1xiv): Estimating the Value of Oil Capital in a Small Open Economy: The Venezuela's Example
NRM	107 2003	Anne Sophie CRÉPIN (lxiv): Threshold Effects in Coral Reef Fisheries
NRM	106.2003	Anne Sophie CRÉPIN (Ixiv): Management Challenges for Multiple-Species Boreal Forests
CLIM	105.2003	Anil MARKANDYA and Dirk T.G. RÜBBELKE: Ancillary Benefits of Climate Policy
		International Climate Agreements
CLIM	104.2003	Barbara BUCHNER and Carlo CARRARO: Emissions Trading Regimes and Incentives to Participate in
CLIM	103.2003	Barbara BUCHNER and Carlo CARRARO: China and the Evolution of the Present Climate Regime
		Island of Montreal: Tensions Between Two Majority Groups in a Multicultural City
KNOW	102.2003	Sébastien ARCAND, Danielle JUTEAU, Sirma BILGE, and Francine LEMIRE (lxiii) : Municipal Reform on the
KNOW	101.2003	David MAY (lxiii): The Struggle of Becoming Established in a Deprived Inner-City Neighbourhood
KNOW	100.2003	Alaknanda PATEL (lxiii): Cultural Diversity and Conflict in Multicultural Cities
KNOW	99.2003	Richard THOMPSON FORD (lxiii): Cultural Rights and Civic Virtue
		Schemes to Socio-Political Realities
KNOW	98.2003	John CROWLEY, Marie-Cecile NAVES (lxiii): Anti-Racist Policies in France. From Ideological and Historical
		Matters
CTN	97.2003	Steven J. BRAMS, Michael A. JONES, and D. Marc KILGOUR: Forming Stable Coalitions: The Process
		Oil Stock Prices
IEM	96.2003	Alessandro LANZA, Matteo MANERA, Margherita GRASSO and Massimo GIOVANNINI: Long-run Models of
		Using Principal Components
IEM	95.2003	Matteo MANERA and Angelo MARZULLO: Modelling the Load Curve of Aggregate Electricity Consumption
CTN	94.2003	Parkash CHANDER: The γ -Core and Coalition Formation
ETA	93.2003	Andrea BELTRATTI: Socially Responsible Investment in General Equilibrium
		Policy: The Case of Russia
IEM	92.2003	A. MARKANDYA, A. GOLUB and E. STRUKOVA: The Influence of Climate Change Considerations on Energy
CLIM	91.2003	Marzio GALEOITI and Barbara BUCHNER: Climate Policy and Economic Growth in Developing Countries
CLIM	90.2003	Marzio GALEOITI: Environment and Economic Growth: Is Technical Change the Key to Decoupling?
CLIM	89.2003	Marzio GALEOITI: Economic Development and Environmental Protection
or n -	00.000	<u>Global Warming</u>
CLIM	88.2003	Johan EYCKMANS and Michael FINUS: New Roads to International Environmental Agreements: The Case of
~ ~ ~		How resource prices affect long-term R&D investments
SIEV	87.2003	Lucas BRETSCGHER and Sjak SMULDERS: Sustainability and Substitution of Exhaustible Natural Resources.
		for the Mezzogiorno
KNOW	86.2003	Elena BELLINI, Gianmarco I.P. OTTAVIANO and Dino PINELLI: The ICT Revolution: opportunities and risks
		The cross-country evidence
NRM	85.2003	Rinaldo BRAU, Alessandro LANZA and Francesco PIGLIARU: <u>How Fast are the Tourism Countries Growing?</u>
CLIM	84.2003	Reyer GERLAGH and Wietze LISE: Induced Technological Change Under Carbon Taxes
CLIM	83.2003	Giuseppe DI VITA: Is the Discount Rate Relevant in Explaining the Environmental Kuznets Curve?
		Environmental Regulation
CLIM	82.2003	Y. Hossein FARZIN and Jinhua ZHAO: Pollution Abatement Investment When Firms Lobby Against
		Petroleum Markets
IEM	81.2003	Alessandro LANZA, Matteo MANERA and Massimo GIOVANNINI: Oil and Product Dynamics in International

NOTE DI LAVORO PUBLISHED IN 2004

IEM	1.2004	Anil MARKANDYA, Suzette PEDROSO and Alexander GOLUB: Empirical Analysis of National Income and So2 Emissions in Selected European Countries
ETA	2.2004	Masahisa FUJITA and Shlomo WEBER: Strategic Immigration Policies and Welfare in Heterogeneous Countries
PRA	3.2004	Adolfo DI CARLUCCIO. Giovanni FERRI. Cecilia FRALE and Ottavio RICCHI: Do Privatizations Boost
		Household Shareholding? Evidence from Italy
ETA	4.2004	Victor GINSBURGH and Shlomo WEBER: Languages Disenfranchisement in the European Union
ETA	5.2004	Romano PIRAS: Growth, Congestion of Public Goods, and Second-Best Optimal Policy
CCMP	6.2004	Herman R.J. VOLLEBERGH: Lessons from the Polder: Is Dutch CO2-Taxation Optimal
PRA	7.2004	Sandro BRUSCO, Giuseppe LOPOMO and S. VISWANATHAN (lxv): Merger Mechanisms
PKA	8.2004	Wolfgang AUSSENEGG, Pegaret PICHLER and Alex STOMPER (IXV): <u>IPO Pricing with Bookbuilding, and a</u> When Issued Market
PRA	9.2004	Pegaret PICHLER and Alex STOMPER (lxv): Primary Market Design: Direct Mechanisms and Markets
PRA	10.2004	Florian ENGLMAIER, Pablo GUILLEN, Loreto LLORENTE, Sander ONDERSTAL and Rupert SAUSGRUBER
		(lxv): The Chopstick Auction: A Study of the Exposure Problem in Multi-Unit Auctions
PRA	11.2004	<i>Bjarne BRENDSTRUP and Harry J. PAARSCH</i> (lxv): <u>Nonparametric Identification and Estimation of Multi-</u> Unit, Sequential, Oral, Ascending-Price Auctions With Asymmetric Bidders
PRA	12.2004	Ohad KADAN (lxv): Equilibrium in the Two Player, k-Double Auction with Affiliated Private Values
PRA	13.2004	Maarten C.W. JANSSEN (lxv): Auctions as Coordination Devices
PRA	14.2004	Gadi FIBICH, Arieh GAVIOUS and Aner SELA (lxv): All-Pay Auctions with Weakly Risk-Averse Buyers
PRA	15.2004	Orly SADE, Charles SCHNITZLEIN and Jaime F. ZENDER (Ixv): <u>Competition and Cooperation in Divisible</u> <u>Good Auctions: An Experimental Examination</u>
PRA	16.2004	Marta STRYSZOWSKA (lxv): Late and Multiple Bidding in Competing Second Price Internet Auctions
CCMP	17.2004	Slim Ben YOUSSEF: <u>R&D in Cleaner Technology and International Trade</u>
NRM	18.2004	Angelo ANTOCI, Simone BORGHESI and Paolo RUSSU (Ixvi): <u>Biodiversity and Economic Growth:</u>
SIEV	10 2004	Stabilization Versus Preservation of the Ecological Dynamics
SIEV	19.2004	Pay in a Contingent Valuation Study: The Value of S. Frasmo in the Lagoon of Venice
NRM	20.2004	Guido CANDELA and Roberto CELLINI (lxvii): Investment in Tourism Market: A Dynamic Model of
		Differentiated Oligopoly
NRM	21.2004	Jacqueline M. HAMILTON (lxvii): Climate and the Destination Choice of German Tourists
NRM	22.2004	Javier Rey-MAQUIEIRA PALMER, Javier LOZANO IBÁÑEZ and Carlos Mario GÓMEZ GÓMEZ (lxvii):
		Land, Environmental Externalities and Tourism Development
NRM	23.2004	Pius ODUNGA and Henk FOLMER (lxvii): Profiling Tourists for Balanced Utilization of Tourism-Based Resources in Kenya
NRM	24.2004	Jean-Jacques NOWAK, Mondher SAHLI and Pasquale M. SGRO (Ixvii): Tourism, Trade and Domestic Welfare
NRM	25.2004	Riaz SHAREEF (lxvii): Country Risk Ratings of Small Island Tourism Economies
NRM	26.2004	Juan Luis EUGENIO-MARTÍN, Noelia MARTÍN MORALES and Riccardo SCARPA (Ixvii): Tourism and
		Economic Growth in Latin American Countries: A Panel Data Approach
NRM	27.2004	Raúl Hernández MARTÍN (lxvii): Impact of Tourism Consumption on GDP. The Role of Imports
CSRM	28.2004	Nicoletta FERRO: Cross-Country Ethical Dilemmas in Business: A Descriptive Framework
NRM	29.2004	<i>Marian WEBER</i> (lxvi): <u>Assessing the Effectiveness of Tradable Landuse Rights for Biodiversity Conservation:</u> an Application to Canada's Boreal Mixedwood Forest
NRM	30.2004	Trond BJORNDAL, Phoebe KOUNDOURI and Sean PASCOE (lxvi): Output Substitution in Multi-Species
		Trawl Fisheries: Implications for Quota Setting
CCMP	31.2004	Marzio GALEOTTI, Alessandra GORIA, Paolo MOMBRINI and Evi SPANTIDAKI: Weather Impacts on
~ ~ ~	22 2004	Natural, Social and Economic Systems (WISE) Part I: Sectoral Analysis of Climate Impacts in Italy
ССМР	32.2004	Marzio GALEOTTI, Alessanara GORIA, Paolo MOMBRINI and EVI SPANTIDARI. weather impacts on Natural Social and Economic Systems (WISE) Part II: Individual Percention of Climate Extremes in Italy
CTN	33 2004	Wilson PEREZ: Divide and Conquer: Noisy Communication in Networks, Power, and Wealth Distribution
VTHC	34 2004	Gianmarco I.P. OTTAVIANO and Giovanni PERI (Ixviji): The Economic Value of Cultural Diversity: Evidence
KIIIC	51.2001	from US Cities
KTHC	35.2004	Linda CHAIB (Ixviii): Immigration and Local Urban Participatory Democracy: A Boston-Paris Comparison
KTHC	36.2004	Franca ECKERT COEN and Claudio ROSSI (Ixviii): Foreigners, Immigrants, Host Cities: The Policies of
		Multi-Ethnicity in Rome. Reading Governance in a Local Context
KTHC	37.2004	Kristine CRANE (lxviii): Governing Migration: Immigrant Groups' Strategies in Three Italian Cities – Rome,
	20 200 4	Naples and Bari
KTHC	38.2004	<i>Kijiemariam HAMDE</i> (IXVIII): <u>Mind in Africa, Body in Europe: The Struggle for Maintaining and Transforming</u>
ETA	39 2004	Alberto CAVALIERE: Price Competition with Information Disparities in a Vertically Differentiated Duopoly
	40 2004	Andrea RIGANO and Stef PROOST: The Opening of the European Electricity Market and Environmental
ґКА	10.2004	Policy: Does the Degree of Competition Matter?
CCMP	41.2004	Micheal FINUS (lxix): International Cooperation to Resolve International Pollution Problems
KTHC	42.2004	Francesco CRESPI: Notes on the Determinants of Innovation: A Multi-Perspective Analysis

CTN	43.2004	Sergio CURRARINI and Marco MARINI: Coalition Formation in Games without Synergies
CTN	44.2004	Marc ESCRIHUELA-VILLAR: Cartel Sustainability and Cartel Stability
NRM	45.2004	Sebastian BERVOETS and Nicolas GRAVEL (lxvi): <u>Appraising Diversity with an Ordinal Notion of Similarity</u> : An Axiomatic Approach
NRM	46.2004	Signe ANTHON and Bo JELLESMARK THORSEN (lxvi): Optimal Afforestation Contracts with Asymmetric Information on Private Environmental Benefits
NRM	47.2004	John MBURU (lxvi): Wildlife Conservation and Management in Kenya: Towards a Co-management Approach
NRM	48.2004	<i>Ekin BIROL, Ágnes GYOVAI and Melinda SMALE</i> (lxvi): <u>Using a Choice Experiment to Value Agricultural</u> Biodiversity on Hungarian Small Farme: Agri Environmental Policies in a Transition al Economy
CCMP	49.2004	Gernot KLEPPER and Sonja PETERSON: The EU Emissions Trading Scheme. Allowance Prices, Trade Flows, Competitiveness Effects
GG	50.2004	Scott BARRETT and Michael HOEL: Optimal Disease Eradication
CTN	51.2004	Dinko DIMITROV, Peter BORM, Ruud HENDRICKX and Shao CHIN SUNG: Simple Priorities and Core Stability in Hedonic Games
SIEV	52.2004	Francesco RICCI: Channels of Transmission of Environmental Policy to Economic Growth: A Survey of the Theory
SIEV	53.2004	Anna ALBERINI, Maureen CROPPER, Alan KRUPNICK and Nathalie B. SIMON: <u>Willingness to Pay for</u> Mortality Risk Reductions: Does Latency Matter?
NRM	54.2004	Ingo BRÄUER and Rainer MARGGRAF (lxvi): Valuation of Ecosystem Services Provided by Biodiversity Conservation: An Integrated Hydrological and Economic Model to Value the Enhanced Nitrogen Retention in Renaturated Streams
NRM	55.2004	Timo GOESCHL and Tun LIN (lxvi): Biodiversity Conservation on Private Lands: Information Problems and Regulatory Choices
NRM CCMP	56.2004 57.2004	<i>Tom DEDEURWAERDERE</i> (lxvi): <u>Bioprospection: From the Economics of Contracts to Reflexive Governance</u> <i>Katrin REHDANZ and David MADDISON:</i> The Amenity Value of Climate to German Households
CCMP	58 2004	Koon SMEKENS and Rob VAN DER ZWAAN: Environmental Externalities of Geological Carbon Sequestration
NDM	50 2004	Effects on Energy Scenarios
NKM	59.2004	Analysis to Evaluate Environmentally Conscious Tourism Management
NRM	60.2004	Timo GOESCHL and Danilo CAMARGO IGLIORI (Ixvi): Property Rights Conservation and Development: An Analysis of Extractive Reserves in the Brazilian Amazon
ССМР	61.2004	Barbara BUCHNER and Carlo CARRARO: Economic and Environmental Effectiveness of a Technology-based Climate Protocol
NRM	62.2004	Elissaios PAPYRAKIS and Reyer GERLAGH: Resource-Abundance and Economic Growth in the U.S.
NRM	63.2004	<i>Györgyi BELA, György PATAKI, Melinda SMALE and Mariann HAJDU</i> (lxvi): <u>Conserving Crop Genetic</u> <u>Resources on Smallholder Farms in Hungary: Institutional Analysis</u>
NRM	64.2004	E.C.M. RUIJGROK and E.E.M. NILLESEN (lxvi): The Socio-Economic Value of Natural Riverbanks in the Netherlands
NRM	65.2004	<i>E.C.M. RUIJGROK</i> (lxvi): <u>Reducing Acidification: The Benefits of Increased Nature Quality. Investigating the</u> Possibilities of the Contingent Valuation Method
ETA	66.2004	Giannis VARDAS and Anastasios XEPAPADEAS: Uncertainty Aversion, Robust Control and Asset Holdings
GG	67.2004	Anastasios XEPAPADEAS and Constadina PASSA: <u>Participation in and Compliance with Public Voluntary</u> Environmental Programs: An Evolutionary Approach
GG	68.2004	Michael FINUS: Modesty Pays: Sometimes!
NRM	69.2004	Trond BJØRNDAL and Ana BRASÃO: The Northern Atlantic Bluefin Tuna Fisheries: Management and Policy Implications
CTN	70.2004	Alejandro CAPARRÓS, Abdelhakim HAMMOUDI and Tarik TAZDAÏT: On Coalition Formation with Heterogeneous Agents
IEM	71.2004	Massimo GIOVANNINI, Margherita GRASSO, Alessandro LANZA and Matteo MANERA: <u>Conditional</u> Correlations in the Returns on Oil Companies Stock Prices and Their Determinants
IEM	72.2004	Alessandro LANZA, Matteo MANERA and Michael MCALEER: Modelling Dynamic Conditional Correlations
SIEV	73.2004	Margarita GENIUS and Elisabetta STRAZZERA: The Copula Approach to Sample Selection Modelling:
CCMP	74.2004	<i>Rob DELLINK and Ekko van IERLAND</i> : <u>Pollution Abatement in the Netherlands: A Dynamic Applied General</u>
ETA	75.2004	Rosella LEVAGGI and Michele MORETTO: Investment in Hospital Care Technology under Different Purchasing Rules: A Real Option Approach
CTN	76.2004	Salvador BARBERÀ and Matthew O. JACKSON (lxx): On the Weights of Nations: Assigning Voting Weights in Heterogeneous Union
CTN	77.2004	Àlex ARENAS, Antonio CABRALES, Albert DÍAZ-GUILERA, Roger GUIMERÀ and Fernando VEGA-
OTM	70 000 4	REDONDO (lxx): Optimal Information Transmission in Organizations: Search and Congestion
CTN	78.2004 79.2004	<i>Francis BLOCH and Armando GOMES</i> (Ixx): <u>Contracting with Externalities and Outside Options</u> <i>Rabah AMIR, Effrosyni DIAMANTOUDI and Licun XUE</i> (Ixx): <u>Merger Performance under Uncertain Efficiency</u>
CTN	00 200 4	Gains
CTN	80.2004 81.2004	Daniel DIERMEIER, Hülya ERASLAN and Antonio MERLO (lxx): <u>Bicameralism and Government Formation</u>

CTN	82.2004	Rod GARRATT, James E. PARCO, Cheng-ZHONG QIN and Amnon RAPOPORT (lxx): Potential Maximization
		and Coalition Government Formation
CTN	83.2004	Kfir ELIAZ, Debraj RAY and Ronny RAZIN (lxx): Group Decision-Making in the Shadow of Disagreement
CTN	84.2004	Sanjeev GOYAL, Marco van der LEIJ and José Luis MORAGA-GONZÀLEZ (lxx): <u>Economics: An Emerging</u>
		Small World?
CTN	85.2004	Edward CARTWRIGHT (lxx): Learning to Play Approximate Nash Equilibria in Games with Many Players
IEM	86.2004	Finn R. FØRSUND and Michael HOEL: Properties of a Non-Competitive Electricity Market Dominated by
		<u>Hydroelectric Power</u>
KTHC	87.2004	Elissaios PAPYRAKIS and Reyer GERLAGH: Natural Resources, Investment and Long-Term Income
CCMP	88.2004	Marzio GALEOTTI and Claudia KEMFERT: Interactions between Climate and Trade Policies: A Survey
IEM	89.2004	A. MARKANDYA, S. PEDROSO and D. STREIMIKIENE: Energy Efficiency in Transition Economies: Is There
~~		Convergence Towards the EU Average?
GG	90.2004	Rolf GOLOMBEK and Michael HOEL: <u>Climate Agreements and Technology Policy</u>
PKA	91.2004	Sergei IZMALKOV (Ixv): <u>Multi-Unit Open Ascending Price Efficient Auction</u>
KIHC	92.2004	Gianmarco I.P. OTTAVIANO and Giovanni PERT. <u>Cities and Cunures</u>
KTHC	93.2004	Massimo DEL GATTO: Agglomeration, Integration, and Territorial Authority Scale in a System of Trading
	04 000 4	<u>Cities. Centralisation versus devolution</u>
CCMP	94.2004	Pierre-Andre JOUVET, Philippe MICHEL and Gilles ROTILLON: Equilibrium with a Market of Permits
CCMP	95.2004	Bob van der ZWAAN and Reyer GERLAGH: Climate Uncertainty and the Necessity to Transform Global
CCMD	06 2004	Energy Supply
CCMP	96.2004	the Implications of Climate Change: See Level Pice
CTN	97 2004	Custavo RERGANTIÑOS and Juan I VIDAL-PUGA: Defining Rules in Cost Spanning Tree Problems Through
CIN	97.2004	the Canonical Form
CTN	98 2004	Siddhartha RANDYOPADHYAY and Mandar OAK: Party Formation and Coalitional Bargaining in a Model of
CIN	70.2004	Pronortional Representation
GG	99 2004	Hans-Peter WEIKARD, Michael FINUS and Juan-Carlos ALTAMIRANO-CABRERA: The Impact of Surplus
00	<i>,,,,</i> ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	Sharing on the Stability of International Climate Agreements
SIEV	100.2004	<i>Chiara M. TRAVISI and Peter NIJKAMP</i> : Willingness to Pay for Agricultural Environmental Safety: Evidence
		from a Survey of Milan, Italy, Residents
SIEV	101.2004	Chiara M. TRAVISI, Raymond J. G. M. FLORAX and Peter NIJKAMP: A Meta-Analysis of the Willingness to
		Pay for Reductions in Pesticide Risk Exposure
NRM	102.2004	Valentina BOSETTI and David TOMBERLIN: Real Options Analysis of Fishing Fleet Dynamics: A Test
CCMP	103.2004	Alessandra GORIA e Gretel GAMBARELLI: Economic Evaluation of Climate Change Impacts and Adaptability
		<u>in Italy</u>
PRA	104.2004	Massimo FLORIO and Mara GRASSENI: The Missing Shock: The Macroeconomic Impact of British
		Privatisation
PRA	105.2004	John BENNETT, Saul ESTRIN, James MAW and Giovanni URGA: Privatisation Methods and Economic Growth
		in Transition Economies
PRA	106.2004	Kira BORNER: The Political Economy of Privatization: Why Do Governments Want Reforms?
PKA	107.2004	Pehr-Johan NORBACK and Lars PERSSON: Privatization and Restructuring in Concentrated Markets
SIEV	108.2004	Angela GRANZOTTO, Fabio PRANOVI, Simone LIBRALATO, Patrizia TORRICELLI and Danilo
		MAINARDI: Comparison between Artisanal Fishery and Manila Clam Harvesting in the vehice Lagoon by
CTN	100 2004	<u>Using Ecosystem Indicators: An Ecological Economics Perspective</u>
CIN	109.2004	Somdeb LARIKI. The Cooperative Theory of Two Sided Matching Problems. A Re-examination of Some
NRM	110 2004	<u>Giusenne DI VIT4</u> : Natural Resources Dynamics: Another Look
SIEV	111 2004	Anna ALBERINI Alistair HUNT and Anil MARKANDYA. Willingness to Pay to Reduce Mortality Risks:
	111.2001	Evidence from a Three-Country Contingent Valuation Study
KTHC	112.2004	Valeria PAPPONETTI and Dino PINELLI: Scientific Advice to Public Policy-Making
SIEV	113.2004	Paulo A.L.D. NUNES and Laura ONOFRI: The Economics of Warm Glow: A Note on Consumer's Behavior
		and Public Policy Implications
IEM	114.2004	Patrick CAYRADE: Investments in Gas Pipelines and Liquefied Natural Gas Infrastructure What is the Impact
		on the Security of Supply?
IEM	115.2004	Valeria COSTANTINI and Francesco GRACCEVA: Oil Security. Short- and Long-Term Policies
IEM	116.2004	Valeria COSTANTINI and Francesco GRACCEVA: Social Costs of Energy Disruptions
IEM	117.2004	Christian EGENHOFER, Kyriakos GIALOGLOU, Giacomo LUCIANI, Maroeska BOOTS, Martin SCHEEPERS,
		Valeria COSTANTINI, Francesco GRACCEVA, Anil MARKANDYA and Giorgio VICINI: Market-Based Options
	110 000 /	tor Security of Energy Supply
IEM	118.2004	David FISK: <u>Transport Energy Security</u> . The Unseen Risk?
IEM	119.2004	Giacomo LUCIANI: Security of Supply for Natural Gas Markets. What is it and What is it not?
IEM	120.2004	L.J. de VRIES and R.A. HAKVOORT: The Question of Generation Adequacy in Liberalised Electricity Markets
KTHC	121.2004	Alberto PETRUCCI: Asset Accumulation, Fertility Choice and Nondegenerate Dynamics in a Small Open
NDM	122 2004	ECONOMY Cardo CIUDDONI, Janoalan MVSIAK and Anter EASSIO, An Internet J.A. Source of Frances of Co. W. (
INIKIVI	122.2004	Carlo GIUF POIN, Jarosiaw MISIAN and Anila PASSIC: An Integrated Assessment Framework for Water Resources Management: A DSS Tool and a Dilot Study Application
NRM	123 2004	Margaretha RRFII Anita FASSIO Carlo GIUPPONI and Paolo ROSATO: Evoluation of Urban Improvement
1 11/11	123.2004	on the Islands of the Venice Lagoon: A Spatially-Distributed Hedonic-Hierarchical Approach

ETA	124.2004	Paul MENSINK: Instant Efficient Pollution Abatement Under Non-Linear Taxation and Asymmetric
		Information: The Differential Tax Revisited
NRM	125.2004	Mauro FABIANO, Gabriella CAMARSA, Rosanna DURSI, Roberta IVALDI, Valentina MARIN and Francesca
		PALMISANI: Integrated Environmental Study for Beach Management: A Methodological Approach
PRA	126.2004	Irena GROSFELD and Iraj HASHI: The Emergence of Large Shareholders in Mass Privatized Firms: Evidence
		from Poland and the Czech Republic
CCMP	127.2004	Maria BERRITTELLA, Andrea BIGANO, Roberto ROSON and Richard S.J. TOL: <u>A General Equilibrium</u>
		Analysis of Climate Change Impacts on Tourism
CCMP	128.2004	Reyer GERLAGH: A Climate-Change Policy Induced Shift from Innovations in Energy Production to Energy
		Savings
NRM	129.2004	Elissaios PAPYRAKIS and Reyer GERLAGH: Natural Resources, Innovation, and Growth
PRA	130.2004	Bernardo BORTOLOTTI and Mara FACCIO: Reluctant Privatization
SIEV	131.2004	Riccardo SCARPA and Mara THIENE: Destination Choice Models for Rock Climbing in the Northeast Alps: A
		Latent-Class Approach Based on Intensity of Participation
SIEV	132.2004	Riccardo SCARPA Kenneth G. WILLIS and Melinda ACUTT: Comparing Individual-Specific Benefit Estimates
		for Public Goods: Finite Versus Continuous Mixing in Logit Models
IEM	133.2004	Santiago J. RUBIO: On Capturing Oil Rents with a National Excise Tax Revisited
ETA	134.2004	Ascensión ANDINA DÍAZ: Political Competition when Media Create Candidates' Charisma
SIEV	135.2004	Anna ALBERINI: Robustness of VSL Values from Contingent Valuation Surveys
CCMP	136.2004	Gernot KLEPPER and Sonja PETERSON: Marginal Abatement Cost Curves in General Equilibrium: The
		Influence of World Energy Prices
ETA	137.2004	Herbert DAWID, Christophe DEISSENBERG and Pavel ŠEVČIK: Cheap Talk, Gullibility, and Welfare in an
		Environmental Taxation Game
CCMP	138.2004	ZhongXiang ZHANG: The World Bank's Prototype Carbon Fund and China
CCMP	139.2004	Reyer GERLAGH and Marjan W. HOFKES: Time Profile of Climate Change Stabilization Policy
NRM	140.2004	Chiara D'ALPAOS and Michele MORETTO: The Value of Flexibility in the Italian Water Service Sector: A
		Real Option Analysis
PRA	141.2004	Patrick BAJARI, Stephanie HOUGHTON and Steven TADELIS (lxxi): Bidding for Incompete Contracts
PRA	142.2004	Susan ATHEY, Jonathan LEVIN and Enrique SEIRA (lxxi): Comparing Open and Sealed Bid Auctions: Theory
		and Evidence from Timber Auctions
PRA	143.2004	David GOLDREICH (lxxi): Behavioral Biases of Dealers in U.S. Treasury Auctions
PRA	144.2004	Roberto BURGUET (Ixxi): Optimal Procurement Auction for a Buyer with Downward Sloping Demand: More
		Simple Economics
PRA	145.2004	Ali HORTACSU and Samita SAREEN (lxxi): Order Flow and the Formation of Dealer Bids: An Analysis of
		Information and Strategic Behavior in the Government of Canada Securities Auctions

(lix) This paper was presented at the ENGIME Workshop on "Mapping Diversity", Leuven, May 16-17, 2002

(lx) This paper was presented at the EuroConference on "Auctions and Market Design: Theory, Evidence and Applications", organised by the Fondazione Eni Enrico Mattei, Milan, September 26-28, 2002

(lxi) This paper was presented at the Eighth Meeting of the Coalition Theory Network organised by the GREQAM, Aix-en-Provence, France, January 24-25, 2003

(lxii) This paper was presented at the ENGIME Workshop on "Communication across Cultures in Multicultural Cities", The Hague, November 7-8, 2002

(lxiii) This paper was presented at the ENGIME Workshop on "Social dynamics and conflicts in multicultural cities", Milan, March 20-21, 2003

(lxiv) This paper was presented at the International Conference on "Theoretical Topics in Ecological Economics", organised by the Abdus Salam International Centre for Theoretical Physics - ICTP, the Beijer International Institute of Ecological Economics, and Fondazione Eni Enrico Mattei – FEEM Trieste, February 10-21, 2003

(lxv) This paper was presented at the EuroConference on "Auctions and Market Design: Theory, Evidence and Applications" organised by Fondazione Eni Enrico Mattei and sponsored by the EU, Milan, September 25-27, 2003

(lxvi) This paper has been presented at the 4th BioEcon Workshop on "Economic Analysis of Policies for Biodiversity Conservation" organised on behalf of the BIOECON Network by Fondazione Eni Enrico Mattei, Venice International University (VIU) and University College London (UCL), Venice, August 28-29, 2003

(lxvii) This paper has been presented at the international conference on "Tourism and Sustainable Economic Development – Macro and Micro Economic Issues" jointly organised by CRENoS (Università di Cagliari e Sassari, Italy) and Fondazione Eni Enrico Mattei, and supported by the World Bank, Sardinia, September 19-20, 2003

(lxviii) This paper was presented at the ENGIME Workshop on "Governance and Policies in Multicultural Cities", Rome, June 5-6, 2003

(lxix) This paper was presented at the Fourth EEP Plenary Workshop and EEP Conference "The Future of Climate Policy", Cagliari, Italy, 27-28 March 2003

(lxx) This paper was presented at the 9th Coalition Theory Workshop on "Collective Decisions and Institutional Design", organised by the Universitat Autònoma de Barcelona and held in Barcelona, Spain, January 30-31, 2004

(lxxi) This paper was presented at the EuroConference on "Auctions and Market Design: Theory, Evidence and Applications", organised by Fondazione Eni Enrico Mattei and Consip and sponsored by the EU, Rome, September 23-25, 2004

2003 SERIES		
CLIM	Climate Change Modelling and Policy (Editor: Marzio Galeotti)	
GG	Global Governance (Editor: Carlo Carraro)	
SIEV	Sustainability Indicators and Environmental Valuation (Editor: Anna Alberini)	
NRM	Natural Resources Management (Editor: Carlo Giupponi)	
KNOW	Knowledge, Technology, Human Capital (Editor: Gianmarco Ottaviano)	
IEM	International Energy Markets (Editor: Anil Markandya)	
CSRM	Corporate Social Responsibility and Management (Editor: Sabina Ratti)	
PRIV	Privatisation, Regulation, Antitrust (Editor: Bernardo Bortolotti)	
ЕТА	Economic Theory and Applications (Editor: Carlo Carraro)	
CTN	Coalition Theory Network	

2004 SERIES		
CCMP	Climate Change Modelling and Policy (Editor: Marzio Galeotti)	
GG	Global Governance (Editor: Carlo Carraro)	
SIEV	Sustainability Indicators and Environmental Valuation (Editor: Anna Alberini)	
NRM	Natural Resources Management (Editor: Carlo Giupponi)	
КТНС	Knowledge, Technology, Human Capital (Editor: Gianmarco Ottaviano)	
IEM	International Energy Markets (Editor: Anil Markandya)	
CSRM	Corporate Social Responsibility and Management (Editor: Sabina Ratti)	
PRA	Privatisation, Regulation, Antitrust (Editor: Bernardo Bortolotti)	
ЕТА	Economic Theory and Applications (Editor: Carlo Carraro)	
CTN	Coalition Theory Network	